DOI: 10.1111/cgf.15112 EUROVIS 2024 C. Garth, A. Kerren, and R. Raidou (Guest Editors)

COMPUTER GRAPHICS forum Volume 43 (2024), Number 3 STAR – State of The Art Report

The State of the Art in Visual Analytics for 3D Urban Data

Fabio Miranda¹, Thomas Ortner, Gustavo Moreira¹, Maryam Hosseini², Milena Vuckovic³, Filip Biljecki⁴, Claudio T. Silva⁵, Marcos Lage⁶, Nivan Ferreira⁷

¹Department of Computer Science, University of Illinois Chicago, USA
²City Form Lab, Massachusetts Institute of Technology, USA
³VRVis Zentrum für Virtual Reality und Visualisierung Forschungs-GmbH, Austria
⁴Department of Architecture and Department of Real Estate, National University of Singapore, Singapore
⁵Tandon School of Engineering and Center for Data Science, New York University, USA
⁶Institudo de Computação, Universidade Federal Fluminense, Brazil
⁷Centro de Informática, Universidade Federal de Pernambuco, Brazil

Abstract

Urbanization has amplified the importance of three-dimensional structures in urban environments for a wide range of phenomena that are of significant interest to diverse stakeholders. With the growing availability of 3D urban data, numerous studies have focused on developing visual analysis techniques tailored to the unique characteristics of urban environments. However, incorporating the third dimension into visual analytics introduces additional challenges in designing effective visual tools to tackle urban data's diverse complexities. In this paper, we present a survey on visual analytics of 3D urban data. Our work characterizes published works along three main dimensions (why, what, and how), considering use cases, analysis tasks, data, visualizations, and interactions. We provide a fine-grained categorization of published works from visualization journals and conferences, as well as from a myriad of urban domains, including urban planning, architecture, and engineering. By incorporating perspectives from both urban and visualization experts, we identify literature gaps, motivate visualization researchers to understand challenges and opportunities, and indicate future research directions.

CCS Concepts

• General and reference \rightarrow Surveys and overviews; • Human-centered computing \rightarrow Visualization; • Applied computing \rightarrow Earth and atmospheric sciences; Environmental sciences; Cartography; Architecture (buildings);

1. Introduction

Over the past few decades, regions around the world have undergone a rapid process of urbanization, resulting in the need for cities to densify to meet rising housing demands. In response to this trend and recognizing the multifaceted and dynamic nature of urban environments, there has been a movement from the traditional use of 2D data and representations towards approaches that better acknowledge the three-dimensional aspects of cities. Many phenomena of interest to a variety of stakeholders, such as civil engineers, urban planners, architects, and climate scientists, are inherently three-dimensional, requiring reasoning over the 3D structure of urban environments. These domains have progressively adopted the third dimension in many of their analytical tasks to study and tackle urban problems. These tasks often rely on data that is intrinsic to the physical aspect of cities. The transition to more sustainable environments, energy sources, and technologies has underscored the importance of leveraging this 3D structure in its entirety. In another front, the advancements of technologies such as virtual reality (VR) and augmented reality (AR) have created opportunities for new experiences in exploring 3D urban environments [CWS^{*}17, CQW17, CMF^{*}22].

However, the inclusion of this additional dimension increases the difficulty in addressing the various challenges involved in designing effective GIS and visual analytics tools. These tools require visual strategies to support analysis of the data referent to the city's geometry (i.e., 3D urban data), navigation to learn the structure of the environment and integration of the information from different points of view, while avoiding common problems such as occlusion and higher cognitive load associated with frequent viewpoint changes and lack of sensorial stimuli [ETT07, BC07]. Tackling these challenges can be fundamental to uncovering features valuable for decision-making and problem-solving in several domains.

But how do domain experts analyze 3D urban data? What are the visual analytics techniques and tools being used by practitioners and experts? To answer such questions and evaluate the state of the art in 3D urban data analytics, in this survey, we have reviewed over 20 journals and conferences from 2008 to 2023, including visualization and cross-cutting multi-disciplinary ones. Our focus is on the analysis tasks, visual analytics tools, applications, and vi-

^{© 2024} The Authors. Computer Graphics Forum published by Eurographics - The European Association for Computer Graphics and John Wiley & Sons Ltd.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

sualization techniques that propose to tackle problems and enable tasks involving 3D urban data. Our survey bridges the knowledge gap between different communities, including visualization, urban planning, architecture, and engineering, and helps identify research challenges that can benefit from these multiple perspectives. Given the diversity of use cases, data, and tasks and the lack of empirical studies on the topic, our primary goal with this survey is to inform the visualization community about challenges and opportunities. We believe that this can foster advancements in both theoretical and applied research in this field and generate a set of well-grounded and concrete recommendations in the future; we hope to inform the visualization community of the challenges and opportunities in 3D urban visual analytics, as well as common needs that urban domain experts have. Our contributions are summarized as follows:

- We first establish a common characterization that allows us to organize contributions from a multitude of domains, including visualization, architecture, engineering, and urban planning.
- We introduce a comprehensive survey on 3D urban visual analytics. Inspired by other surveys [DFT19, PSS*21, CD21], we follow a human-centered, interrogative method in which we classify each evaluated work with respect to Munzner's [Mun15] analytical framework of *Why is 3D urban data being analyzed*, *What data is being analyzed*, and *How it is being analyzed and visualized*.
- We report a series of research directions and open problems in 3D urban visual analytics. These include visual metaphors, navigation & occlusion, data and systems integration, evaluation of visual designs, and guided explorations.

Our survey is organized as follows: Section 2 presents a background on 3D urban analytics and data. In Section 3, we review related surveys. In Section 4, we introduce our survey scope and methodology. Section 5 presents an overview of the survey, and Section 6 presents and discusses the types of papers covered in our survey. In Sections 7 and 8, we group and discuss papers according to primary (i.e., **Why**) and data (i.e., **What**) dimensions. In Section 9, we list common tasks across use cases and, in Section 10, we discuss **how** visualization and interaction techniques facilitate these tasks. In Section 11, we discuss the main observations from our survey and we identify challenges and opportunities for visual analytics of 3D urban data. Section 12 concludes our survey. The full list of reviewed papers can be accessed at https://urbantk.org/survey-3d.

2. Background

3D urban analytics. Cities are three-dimensional in nature and hence, should be described and analyzed in three dimensions, considering their structures and urban forms. For instance, the presence of high-rise buildings affects wireless connectivity, while the height of buildings, the density of roads, and the amount of green spaces influence environmental conditions like air quality and temperature [HM17, CW23]. However, in the absence of appropriate methods and techniques to analyze 3D data, urban morphology analysis has adopted various 2D measures to describe the physical form of cities [BC22]. These measures often fail to draw a vivid picture of the reality of urban life, leading to simplified models, calculations,

and, ultimately, policies that lack the precision needed to translate findings into actionable insights [KD21].

The physical form of cities in many parts of the world is largely influenced by a set of regulations known as zoning laws, which dictate, among others, the size, shape, and bulk of the buildings. Although inherently a three-dimensional concept, zoning codes are conventionally enacted without leveraging 3D information. This makes it very difficult to assess the large-scale impact of new regulations on varying aspects of urban life ranging from wind patterns [Lin05, DvdGP*20, BPMW18], air quality [ZXL22, CLJ*19], and heat emission [DYW*19, YAM*20] to the flow of people, goods, and services [LCL22, ZC22]. 3D models provide a more realistic view of these impacts, empowering local communities to understand better the implications of zoning codes on their everyday life [BPMW18].

The use of 3D analysis in urban planning and policy also extends to other areas, such as urban ecology [TZQ*19]. For example, the Urban Heat Island (UHI) effect, which is characterized by a sharp rise in urban temperature, poses a significant threat to public health, the ecological environment, and urban livability [EMM17,HW19b]. 3D models can help mitigate the UHI effect by providing a deeper understanding of factors such as the sky view, building bulk, and horizontal reflectance [CG17,GLU09,BRV*17, GCB20,RcH*22]. Additionally, 3D analysis is crucial in the realm of urban disaster management, enabling the accurate identification of catchment areas in events such as flooding [WHM*19], earthquakes [RTCVBS17], or even the outbreaks of infectious disease [MG20,ETVF20,Els21,XGY*23].

Urban digital twins, which have gained increasing attention in recent years, provide a virtual representation of cities as complex systems [ABB*21, Cha22, LJSB23]. These digital replicas usually employ 3D visualization to bring together stakeholders and various sources of heterogeneous data, models, and algorithms describing various parts of urban systems such as urban ecology [SH20], transportation [JMB*22], and economic and social functions [Bat18, DWL*20, ABB*21]. They enable 3D spatiotemporal simulations and impact assessment analysis. Urban digital twins can inform decision-makers about the multi-scale impacts of policies as well as empower participatory design and community involvement in urban planning [LTM19, DWY19]. While previous works have extensively leveraged 2D screens, VR and AR technologies also have the potential to offer powerful data analysis capabilities while providing an immersive experience [CWS*17]. These systems open avenues for novel interactions to enhance visual data analysis in 3D urban environments, by providing new strategies for city navigation [CMF*22] and to deal with building occlusion [COW17].

3D urban data. Urban visual analytics is composed of a set of stages to process, analyze, and visualize data. In the context of 3D urban data, the first stage involves the collection of 3D data describing the physical layers. This data can originate from a variety of sources, including authoritative city agencies and crowdsourced initiatives like OpenStreetMap [HW08]. The methods employed to acquire this data encompass a range of techniques, such as sensing with LiDAR or mobile mapping. In this survey, we define 3D urban data as the information inherently associated with the three-



Figure 1: Overview of the survey, centered around paper type and three main dimensions asking Why, What, and How visualization supports 3D urban data analytics. Each dimension corresponds to a section of the survey, and major subsections highlight the specific tags considered in each category. We also characterize papers according to their primary display modality and evaluation method. Top left: an illustration of the thematic layer (3D urban data with sunlight access information) and the physical layer (physical form of Downtown New York City).

dimensional structure of urban environments. Depending on the use case, such data can also contain attributes describing specific properties of the physical form of the city. The data is then transformed to satisfy analysis, modeling, or visualization requirements. For example, transforming OpenStreetMap data into polygon meshes and modifying mesh resolutions. Transformed data is optionally stored in general (e.g., MySQL) or 3D-specific (e.g., 3DCityDB) databases to facilitate querying. Finally, the physical layer data is leveraged for analysis and modeling to create new thematic layers, which are then visualized.

Designing visual analytics tools for experts considering 3D urban environments and digital twins poses several challenges along this pipeline. Data acquisition and transformation require the integration and management of disparate data formats (e.g., building footprints, tabular data, heightfields, street networks). Modeling certain phenomena oftentimes requires complex simulations, the involvement of multiple stakeholders, and deep domain expertise. Finally, the analysis and visualization of data in 3D urban scenarios require strategies to map thematic data to the urban environment. This is made more challenging if the data is temporal and/or multivariate. On top of this, tools should also consider strategies to facilitate navigation in the 3D environment and minimize occlusion, both across buildings and within the same building.

While it is also crucial to acknowledge that data visualization in general remains separate from the dimensionality of the data, it is equally important to recognize that visualizing 3D data within a 2D design space can lead to substantial information loss due to its inherent constraints. This stems from overplotting, visual clutter and occlusion. This occurs primarily because numerous data points may be superimposed over each other, making it challenging for a human analyst to discern individual data points and infer potential spatial relationships. Particularly in contexts where depth, spatial relationships and immersion are critical to understanding complex 3D geospatial data and its features, 3D visualizations offer distinct advantages over traditional 2D counterparts.

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd. **Outlook.** In this survey, we organize the targets of analysis into two *layers*. The physical layer represents the physical form of the city and its geometries. The other one called the thematic layer, stores the 3D urban data that is output by simulations, machine learning models, sensing initiatives, or surveys. Figure 1 (top left) illustrates the thematic and physical layers. At the center of our work is the survey of contributions that leveraged these layers in their studies.

3. Related surveys

Despite the growing body of literature on 3D urban visual analytics, no comprehensive survey has been published on it. However, there are surveys on related topics. For example, Kraus et al. reviewed immersive analytics and the use of 3D techniques for visualization in general, not focusing on urban data [KFS*22]. Moreover, the study excluded papers where visualizations are inspected on 2D screens (therefore not covering a large body of work that relies on this modality). Chen et al., on the other hand, reviewed the contribution of visualization for urban analytics to derive a design space for immersive urban analytics systems [CWS*17]. Nevertheless, the study does not include important dimensions explored in our survey (e.g., analytical tasks, occlusion handling techniques) and also excludes relevant work done in other research communities. Doraiswamy et al. presented a high-level overview of the challenges of urban data, focusing on 2D-based contributions, with only a brief discussion of the 3D opportunities [DFL*18]. Other urban analytics surveys, such as Deng et al. [DWL*23], Feng et al. [FQY*22], and Zheng et al. [ZWC*16], did not cover 3D urban data and, more importantly, visualizations and tasks specifically aimed at that type of data.

In summary, this paper aims to fill the existing gaps by reviewing the literature on 3D urban data, analytics tasks, and use cases, surveying current visualization techniques in that space, and contributing a classification to allow connections and collaborations between urban and visualization experts.

Table 1: The main venues reviewed in this survey (horizontal line separates the visualization-related ones and domain application venues, with journals shown first). Papers that appeared between 2008 and 2023 were reviewed and the ones related to 3D urban analytics were included in our analysis. The citations in those papers were also used as sources of relevant prior research.

Computer Graphics Forum
Computers & Graphics
IEEE Computer Graphics and Applications
Information Visualization
ACM Trans. on Graphics
IEEE Trans. on Visualization and Computer Graphics
The Visual Computer
Visual Informatics
ACM Conf. on Human Factors in Computing Systems
Eurographics Conference on Visualization
IEEE Pacific Visualization Symposium
ACM SIG on Comp. Graphics and Inter. Techniques
IEEE Visualization Conference
Building and Environment
Computers, Environment and Urban Systems
Env. and Planning B: Urban Analytics and City Science
Int. Journal of Architectural Computing
Int. Journal of Geographical Information Science
Journal of Photogrammetry and Remote Sensing
ISPRS Ann. of the Phot. Rem. Sens. and Spat. Inf. Sci.
Journal of Urban Design
Landscape and Urban Planning
Sustainable Cities and Society
Symp. on Sim. for Architecture and Urban Design
Urban Climate

4. Survey scope & methodology

For our survey, we have focused our efforts on the selection of papers that (1) made visualization contributions leveraging 3D urban data or facilitating 3D urban visual analytics or (2) made domain-specific contributions generating or analyzing 3D urban data. Hence, our survey covers not only visual analytics tools, visualization techniques, and design and evaluation studies but also domain-specific papers that either analyze or generate 3D urban data. We have focused on the generation of thematic data that is either inherently 3D (e.g., CFD simulation of wind) or derived from the 3D physical layer (e.g., sunlight access). We have deliberately excluded the generation or acquisition of data for the physical layer as these works typically do not contain analytical components or thematic layers which are essential for 3D urban analytics.

The main goal of our survey was to review active areas of visualization research related to visual analysis of 3D urban data. For this reason, we reviewed papers from both visualization and domain-specific venues published between 2008 and 2023. While we acknowledge that there could be previous works published prior to 2008, focusing on the last 15 years helps manage the scope of the review while including studies that are more technologically and methodologically relevant. On the visualization side, we included the main visualization & computer graphics journals (e.g., IEEE Transactions on Visualization and Computer Graphics) and conferences (e.g., IEEE VIS, EuroVis). In domain-specific venues, top journals (e.g., Computers, Environment and Urban Systems,



Figure 2: In the selection, filtering, and tagging steps, each paper was reviewed by at least two of the authors. In the consolidation step, each paper was reviewed by one of the authors. We also highlight the number of papers considered at each step.

Urban Climate) as well as a popular symposium (SimAUD), were included. These venues were selected based on discussions with experts in the field (also co-authors of this survey) and represent the top-ranked journals and conferences in GIS, architecture, remote sensing and urban planning that contain papers related to 3D urban visual/data analytics. Table 1 lists the initial set of visualization and domain-specific venues used in our search.

To select relevant papers for the survey, we followed a four-step approach (Figure 2). In Step 1, we did an exhaustive search through the journals and proceedings highlighted in Table 1. In this step, each author was responsible for reviewing papers between 2008 and 2023 from 3 venues and selecting the ones with components in either (1) 3D urban data or (2) visual analytics for 3D urban data. To achieve this we inspected the title, abstract and figures present in the paper (that could suggest the use and/or analysis of 3D urban data). This step resulted in an initial set of 669 papers. In Step 2, we filtered the papers according to their relevance - i.e., papers with either 3D urban data or visual analytics for 3D urban data. Each paper was randomly assigned to two authors of the survey; each survey author then independently assessed the relevance of the papers assigned to them. For each paper, if two survey authors agreed on the relevance of the paper, the paper was included; if not, the paper was removed. In case of a disagreement, a third survey author made the final decision. In this step, the authors followed a similar procedure as in the previous step, going through the title, abstract, and figures of each paper. A total of 397 papers remained after this step.

In Step 3, we held meetings to discuss a set of informative tags based on the previously identified papers. Our intention was to have tags that reasonably describe each paper along three primary dimensions (*why*, *what*, and *how*), covering use cases, analysis tasks, data, visualization, and interaction. Each paper was then randomly assigned to two authors of the survey, who were responsible for individually classifying the paper using the tags. Since the process of tagging demanded a more careful review of the papers, survey authors could also mark a paper as being outside of the scope of the survey. Following the same procedure outlined in Step 2, if there was disagreement regarding the removal of a paper, a third survey author made the final decision. At the end of this step, out of 397 papers, 175 remained – the other papers were excluded since, after a more careful inspection, they did not satisfy the aforementioned requirements.

In Step 4, one author was responsible for consolidating the classifications of each paper into a final classification. At the end of this step, we had the final 175 tagged papers, including 54 visualization papers and 121 domain-specific ones. These are works where both



Figure 3: Distribution of tags across Paper type, Why, What and How dimensions.

physical and thematic layers play a key role. In other words, works that leverage 3D geometries representing physical layers and data associated with such geometries defining thematic layers. As such, we have excluded works using urban data where there is no direct association between these layers (such as images) or where there is no clear presence of thematic attributes (such as point clouds). For example, works that primarily focus on *extracting* polygon meshes from images or point clouds are outside the scope of this survey.

Limitations. Given the multidisciplinary field, urban analytics found its way into a wide range of journals and conferences across different fields. Therefore, it is unfeasible for this survey to cover all the possible venues that might publish papers related to urban studies. Hence, early on, we decided to focus on two main types of venues: visualization ones and cross-cutting multidisciplinary venues that have, in their aims, topics related to urban problems and cities. Venues that do not satisfy these requirements include very specific (e.g., Journal of Wind Engineering and Industrial Aerodynamics) to multidisciplinary ones (e.g., Nature). In doing so, we were able to restrict the number of surveyed venues while maintaining a reasonable sample of use cases, analysis workflows, and visualization techniques. We believe that, while not an exhaustive search through every journal and conference that might publish on urban studies, we present a well-defined picture of the topic to bridge the gap between disciplines.

5. Survey overview and organization

In this survey, we summarize previous visualization and domainspecific contributions using an interrogative method that tries to answer: Why is 3D urban data being analyzed, What data is being analyzed, and finally How it is being analyzed. Given the interdisciplinary nature of urban analytics, our goal was to cover both visualization aspects of the surveyed works (to allow us to identify research gaps more easily) as well as urban domain aspects (to facilitate the creation of a common language between visualization researchers and urban experts, beyond siloed collaborations).

Figure 1 presents an overview of the survey, with the three main questions, their categories, and fine-grained tags. A paper

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd. could be assigned to one or more tags within a given category, except for paper type, where each paper is only assigned to one type. We dedicate a separate section to each one of these questions/dimensions. Section 6 presents the considered paper types in this survey; Section 7 then presents the Why dimensions, namely use cases (7.1), analysis actions (7.2) and targets (7.3); Section 8 presents the What dimensions, that is physical data entities (8.1), thematic data origin (8.2) and properties (8.3), and spatial scope (8.4). Then in Section 9, we provide an overview of the most common tags from the previous sections in order to characterize a set of popular tasks across the use cases. Section 10 discusses the How dimensions, which include visualization encodings (10.1), integration of thematic and physical visualizations (10.2), occlusion handling (10.3), navigation (10.4), integration (10.2), display modalities (10.6), and evaluation methods (10.7). Each section starts with a brief overview of that specific dimension and its corresponding categories. Each subsection then drills down into various aspects of that domain, data and visualization & interaction, with plots showing the distribution of tags in that category. For easier referencing, we highlight the first occurrence of the tag in the text. Section 11 presents research directions and opportunities, and Section 12 concludes the survey.

Figure 3 shows a summary of the number of papers with each tag. In the image, tag popularity can be compared across dimensions. Figures 5, 7, 10 provide an overview of the distribution of tags, but now as the intersection across Why, What, and How dimensions. To facilitate the exploration of the distribution considering the intersection of tags, the color of each cell in the matrices is normalized by the total number of papers within each sub-matrix. For example, the color of each cell in the *Use case & Paper type* sub-matrix (Figure 5 (top left)) is normalized by the total number of papers within that sub-matrix.

6. Paper type

One initial effective way to understand the span of contributions in 3D urban data analytics is to classify the research works and their paper types. We followed the categories proposed by Munzner [Mun08] and classified visualization papers using the fol-



Figure 4: Example of papers from each one of the types considered in the survey. Left: papers with visualization contributions (system [DFL*15], technique [CBK*19], design study [WKS*14], evaluation [MFS*23]). Right: papers with domain contributions (data creation [LGL114] and application studies [KBHS18]).

lowing categories: system, technique, design study, and evaluation. We note that Munzner's model paper type was not used, since none of the surveyed visualization papers aligned with its classification. While the previous classification is well suited for visualization and visual analytics works, it does not cover the range of possible contributions in domain-specific papers. To account for these papers, we have added two new paper types: data creation and application studies. The data creation type refers to domain-specific works that present methodologies to derive novel pieces of thematic data that are not available or are very difficult to acquire through other methods. Similarly, the application studies type refers to domain-specific papers presenting detailed analytical studies using 3D urban data. We highlight that for these two paper types, the data visualization component is not a contribution, but rather often a tool to inspect and present their results. For example, while design studies propose visual representations for a particular domain problem, application studies contribute domain-specific analysis of 3D urban data. Figure 4 shows examples of this categorization, with visualization and domain contributions. Each of the surveyed papers was assigned to one of the paper types, resulting in the distribution presented on the top left of Figure 3.

7. Primary dimensions (Why)

As metropolitan areas continue to expand and adopt more complex forms, they face new challenges that are large in scope and scale. Facilitated by the proliferation of new sources of data, urban analytics emerged to address those challenges by probing and analyzing cities. At its core is the need to integrate, model, analyze, and visualize multiple sources of data describing different urban phenomena while taking into account the requirements and expectations of multiple stakeholders, from subject matter experts to concerned citizens. In our primary dimensions, we discuss papers taking into account Why 3D urban data is being analyzed. First, we present the result of the classification of the papers considering their primary domain *use case* (Section 7.1). Then, we review the papers according to their *analysis action* (Section 7.2) and *analysis target* (Section 7.3). Figure 5 shows the distribution of tags according to the Why and How dimensions.

7.1. Use cases

The survey covers a wide breadth of urban use cases, each with different data, analytics, and visualization requirements. We classified each paper into the following use cases: sunlight access, wind, view impact, building energy modeling, disaster management, urban climate, noise, property cadastre, and others. Such classification was decided after our initial selection process (detailed in Section 4) and it comprehensively covers all surveyed works. Use cases with fewer than five papers are classified as "others." To facilitate the discussion in this section, the use cases are grouped into three broader themes: (1) contributions that model or analyze the interplay between an inherently 3D natural phenomenon and the built environment (Section 7.1.1); (2) contributions that model or analyze 3D phenomena that are primarily driven by human factors (Section 7.1.2); and (3) contributions that are in and of itself defined by the built environment (Section 7.1.3).

7.1.1. 3D natural phenomenon and the built environment

In the first theme, we have papers that cover sunlight access, wind & ventilation, damage & disaster management, and urban climate. Sunlight access analysis studies the impact of the built environment (primarily buildings) on the equitable "right to light" or "right to sunshine", with a wide range of impacts, from vegetation growth to solar energy potential and building energy consumption. The majority of surveyed papers related to sunlight access were domain-specific ones (25 out of 39), either proposing new efficient approaches to compute data (Figure 4 (Data creation)), studies analyzing the energy potential in different cities [Zha13, BMW*16, DLD*16, KKO18, DLV17, OS18, ZYS*19, FPP*21] or for different scenarios, such as pedestrian comfort [DL19], photovoltaic panels [FWC22, ZWK*22, ZWL*23], urban farming [PSTB21], retrofit [SCF20] or data standards [BLSV16]. While not their primary target, a number of domain-specific papers use sunlight access and shading as one of their attributes of interest [AAA*17, YNK*18, WDDH19, WM20]. Regarding visualization contributions tackling sunlight access, the majority were interactive visualization systems specifically designed for this type of problem [MDL*19] or that use sunlight as an attribute of interest [WHG10, EPTD12, WSK18, ZZL21]. Doraiswamy et al., for instance, present Catalogue, a system that combines different view measures to create building design options (Figure 4 (System)) [DFL*15]. Shadow Profiler is a tool aimed at urban planners and architects driven by the need to assess shadows at a city scale (Figure 8 (left)) [MDL*19]. Technique contributions directly linked with sunlight include Muñoz et al. with real-time computation of solar exposure [MnPBM*17]; Bremer et al. propose a sampling and texture mapping approach for solar potential sim-



Figure 5: Distribution of surveyed papers according to Why and How dimensions, with shades denoting tag occurrence.

ulations [BMW*16]. Indirectly, Pasewaldt et al. use sunlight access as a case for their multi-perspective detail+overview contribution [PTD13]; Vanegas et al. propose an inverse design procedure for quick creation of 3D models [VGDA*12]; and Lorenz and Döllner use sunlight access as a case for their discussion on mapping 3D building geometry to 2D rasters [LD10]. Lastly, Herbert and Chen [HC15] and Mota et al. present evaluation studies, respectively comparing 2D and 3D representations for urban planning and 3D-oriented visualizations for urban analytics (Figure 4 (Evaluation)) [MFS*23].

All of the surveyed papers about wind & ventilation were domain-specific ones (43). Wilkinson et al. present a data generation technique to approximate complex computational fluid dynamics (CFD) using an artificial neural network [WBH14]. The other surveyed papers were primarily analysis contributions that relayed on CFD for building-level [AKH19] or neighborhoodlevel [EGAB17, ZKL*21] wind load analysis (Figure 12 (Superimposition)), wind power potential [SDP20], inner building ventilation [SL19, HP20], wind impact on urban climate [GBI14, DL19, HP21], pollution dispersion [GL19, ZWLZ22, HZL*23, LMS*23], or the interplay between sunlight access and wind for thermal comfort [JY20].

Damage & disaster management was a use case where we had more visualization contributions than domain-specific ones (9 out of 13). Here, we included papers that tackled heavy rain and floods, seismic response, and search & rescue. Analysis papers include Xiong et al. [XLH*15] and Willemborg et al. [WSK18] with applications of 3D urban models on building seismic response

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd. and simulation of detonations. When considering the visualization papers, two present design studies targeting visual analytics for flood management [WKS*14,KWS*14] (Figure 4 (Design study)). Cornel et al. present a technique for surface reconstruction with real-world applications in flood and heavy rain scenarios (Figure 4 (Technique)) [CBK*19]. Other surveyed papers propose visual analytics systems [RWF*13, CWSC14, CKS*15, BSK*17, VSOC21, LLT*19, BKH*23]. Vuckovic et al., for instance, explore the combination of 2D and 3D visualizations for flood and stormwater management [VSOC21]. At the intersection between several use cases (i.e., sunlight access, wind & ventilation, and damage & disaster management use cases), Elsayed investigates the interplay between multiple 3D variables, such as solar radiation, ventilation, and air temperature, for pandemic mitigation plans [Els21].

A large number of papers in the category of 3D natural phenomena primarily tackle problems related to urban climate (30 in total). Most papers (28) were application studies, investigating surface temperature [HW19a], and the impact of urban morphology [KKK21] and redevelopment [KBHS18] on climate (Figure 4 (Application studies)). Other contributions include analysis based on WRF simulations [KPMB21, WHN*21, HYF*23]. Visualization contributions include Gautier et al.'s system to enable domain scientists to analyze the interplay between air temperature and the built environment [GBC20], and Deng et al.'s context-aware technique to visualize street-level data [DZMQ16].

7.1.2. 3D phenomena driven by human factors

In this theme, we have contributions that model or analyze 3D phenomena that are primarily driven by human factors, such as energy modeling or energy potential assessment, noise & sound propagation, and property cadastre. Urban building energy modeling (UBEM) has been recognized as an increasingly useful data-driven approach to tackle the many challenges facing cities today and identify pathways for building retrofit and deployment of city-level renewable energy initiatives [AA19, JPS*20, ABLD*22]. Out of the 23 papers on the topic, 17 were classified as application study papers, either analyzing specific urban cases [OS18, FWC22, ZWK*22], the interplay between sunlight and energy usage [SCF20], or urban climate and energy usage [GBI14, NZCE17, WSK18, HP21]. Other domain-specific contributions include general frameworks (with simple visualization components) for the assessment of energy usage [MTML18, AAA*19, ABLD*22]. Data creation contributions include Albeaik et al. with the creation of 3D urban models for energy and sunlight assessment [AAA*17], and Krietemeyer and Kontar [KK19] and Wolosiuk and Mahdavi [WM20] with data integration methods for building performance assessment. Bartosh and Gu presented an immersive visualization system that enabled users to explore energy consumption data [BG19].

Noise & sound propagation is another use case of interest when considering 3D urban data. In recent decades, noise has become a growing urban problem, impacting public health [HSN14], social well-being [GCA06] and quality of life [DZD*10], as noise increases stress, sleep disruption, annoyance and distraction [BVR07,HDVT*08]. Noise codes typically impose regulations that aim to mitigate the noise at the source (e.g., by erecting sound barriers or modifying building designs) [BH10, HSN14], adding new requirements for urban planners and architects to optimize designs, which in turn creates the need to analyze noise and sound propagation. Out of the eight surveyed works on this topic, six were domain-specific, with analyses exploring the relationship between the built environment and noise [SKK08,ZLH*17,TLH*22]. The only visualization contributions were an immersive system for the visualization of traffic noise [BG19], and a visualization design for continuous noise phenomena, with a case study aimed at urban noise [BJK*22].

Visualizations in 3D also play an important role in the property cadastre case, especially in the last few years with the increase in the demand for 3D real property cadastre [JRK14, DFFL17]. Unlike rural areas, where a 2D description of the land is sufficient, urban areas present several challenges, where property units (and land uses) are positioned on top of each other. In our survey, we reviewed eight papers on this topic. Application studies include Koziatek and Dragićević's iCity3D, a forecasting method for vertical urban development [KD17]. Surveyed visualization papers also include two evaluations on how users perceive transparency [WPH17] and rendering attributes [SAL*20] in visualizations of 3D cadastre; a design study discussing the requirements for 3D cadastre systems [SKB*13]; a visualization prototype for 3D cadastre [SORB18]; a distortion technique for 3D building properties [YCL*19]; and an immersive system that included property management [LLZ*16].

7.1.3. Built environment only

In this theme, we have contributions where the built environment itself is the major focus of analysis. This theme covers view impact analysis, a common operation in architecture and urban planning. Scores computed on the surface of the building summarizing the visibility of certain geographical features (e.g., landmarks, parks, waterfronts). Our survey reviewed 14 papers in this theme. Analysis papers tackled view access equity [YJNY11], visibility to green spaces [YYS*16, VJP*21] and landmarks [SMS23], views of high rises [LXWY22] and office rents [TCFR21]. The visualization papers include: Ortner et al., a design study for visibility-aware urban planning [OSS*17]; Zhang et al., an immersive analytics system that takes into account visibility [ZZL21]; Doraiswamy et al., a topology-based framework for building design [DFL*15]; and Ferreira et al., with a visual analytics system aimed at architects and urban planners that allows for view impact analyses [FLD*15].

Other use cases in this theme include the analysis of urban change over time [BDW*08, KD19], walkability considering 3D footpath networks [SWZ21, ZSW21], simulation of radio propagation in city environments [BCC*18], enclosure assessment [KM17], urban design plans [LS13, Rei15, MKK*18, LSAG22, SD23], and the study of urban vitality [ZY18] (Figure 12 (Linked view)).

7.2. Analysis actions

In order to evaluate the surveyed papers on the reasons *why* they analyze 3D urban data, we follow Brehmer and Munzner and their typology of abstract visualization tasks [BM13, Mun15] that include mid-level (search) and low-level (query) actions. Given that the surveyed works are driven by the discovery and analysis of new information (as opposed to presentation), we decided to focus only on search and query actions. Given the importance of spatial awareness and spatial relation, we incorporate a spatial awareness query action, following Elmqvist and Tsigas' taxonomy of 3D occlusion management for visualization [ET08]. The term *spatial relation* will be used in the remainder of this survey.

7.2.1. Search

Brehmer and Munzner classify search actions according to whether the location and identity of targets of analysis are known to the user. When considering 3D urban data, the physical layer contains the location of the analysis, and the thematic layer contains the identification, i.e., the aspect of the data that the user is interested in. Figure 6 (left) highlights the four possible cases, with a known location represented as a green building **1**, and a known identity represented as a bar chart with selected values **1**. An unknown location is represented by four directional arrows **1**, and an unknown identity is represented by a bar chart without any values selected **1**.

A lookup search is one where the user knows both the location **1** and the identity **11**. For example, a user performing view impact analysis might want to look up the areas of a building with the best views of parks only considering the north-facing façade. Wu et al. present a framework that allows users to perform lookuplike sunlight analysis for a known building at a specific time of the year (Figure 9 (micro)) [WHG10].



Figure 6: Search (left) and query (right) actions.

In a locate search, the user knows the identity he, but does not know the location of the target, requiring them to look around different places of the physical layer \square . For example, a user analyzing UBEM might want to locate the apartment with the highest energy consumption in a certain neighborhood. The domain papers by Yu et al. [YYS*16], and Li et al. [LXWY22] perform a locate search for view impact analysis (Figure 9 (middle, bottom)). In the papers, the analysis is interested in locating places in a façade that have high scores for certain points of interest (e.g., greenery, water body). Locate search has also been highlighted by Huang et al. and Wickramathilaka et al. when visualizing areas exceeding a particular wind speed [HYF*23] and noise [WUAC23] threshold, respectively.

A browse search is one where the user knows the location **\$** but does not know the identity 🕍 of the target. In property cadastre, for example, the user might be interested in searching the number of ownership changes of units in a particular building. As shown in Figure 5, the browse search is the most popular type among the domain papers, which is expected given the non-interactive (i.e., batch-oriented) type of analyses performed in these studies, with limited exploration capabilities. This type of search is also commonly supported by distortion techniques, such as Wu et al. [WP16] and Chen et al. [CMF*22] (Figure 14 (Deformation)), as the user knows the location and distortion is used to enable the identification of the target. The UrbanVR visualization system presented by Zhang et al. [ZZL21] is an example of how the analysis of 3D urban data can be supported by a browse search (Figure 14 (Assisted)): the user is interested in the analysis of different candidate buildings for a particular location, but needs to evaluate against certain attributes (e.g., shading and visibility).

In an explore search, the user does not know the location nor the identity in of the target, usually starting with an overview of the data at a macro scale. In sunlight access analysis, for example, a user might be interested in exploring temporal patterns across all buildings of a neighborhood. This type of search is more commonly supported by visual analytics systems, as shown in Figure 5. Tools like Run Watchers [KWS*14], Urbane [FLD*15], Cornel et al. [CKS*15], Vis-A-Ware [OSS*17], VitalVizor [ZY18], Shadow Profiler [MDL*19], and UrbanVR [ZZL21] all support exploration as part of their analyses. Even though certain domain-specific papers support exploration [YNK*18, TLH*22], they mostly rely on off-the-shelf 3D visualization libraries, with very little effort going towards the customization of these interfaces to satisfy specific tasks. An outlier is the work by Reinhard that combines 2D and 3D visualizations for the exploration of urban planning projects [Rei15].

7.2.2. Query

Following Brehmer and Munzner's query actions, after a target (or set of targets) is found by the user through search actions (either on the physical or thematic layers), the user will then *identify, compare*, or *summarize* these targets. To account for the need to have spatial awareness, we also classify papers according to Elmqvist and Tsigas' idea of *spatial relation*. Figure 6 (right) presents the four possible cases. While an *identify* query refers to a single target **0**, *compare* refers to multiple subsets of targets **10**, and *summarize* will require an overview of a large subset of the data.

In the context of 3D urban data, an identify query is one where the primary target of interest is a single region of the façade, an entire façade, or an entire building. For example, in sunlight access analyses, a user might be interested in identifying floors or windows with the largest amount of accumulated shade over a period of time. Most of the surveyed papers (both domain and visualization ones) support this query. For example, for the identification of radio propagation values [BCC*18], ideal locations for wind turbines in cities [SDP20], discrepancies between simulated and predicted wind values [WBH14], and energy consumption in historical neighborhoods [MTML18].

In a compare query, the user might be interested in comparing different targets – for example, comparing sunlight access of north and south-facing façades or comparing between different *what if* scenarios. Delmas et al. compare different building configurations to maximize solar potential in a neighborhood [DLD*16]. Figure 9 presents two analysis cases where comparisons are at the center of the analysis workflow [LXWY22, YYS*16]. In Figure 9 (middle), the analysis compares two different view impact attributes across buildings in a neighborhood (middle-center) and across a façade (middle-right). Similarly, in Figure 9 (bottom-right), the view access to greenery is compared across three different sides of the building, while (bottom-center) performs a *what-if* analysis (or scenario planning) comparing view access before and after the placement of a tall neighboring building.

In a summarize query, the user is interested in analyzing a large set of targets – for example, an overview of the sunlight access situation across buildings, parks, roads, and sidewalks in a dense neighborhood. Stoter et al. present an overview of the noise impact on a 3D urban environment [SKK08], highlighting limitations of 2D noise maps, and De Luca summarizes wind flows for the assessment of pedestrian comfort at a neighborhood scale [DL19].

Lastly, in a spatial relation query, the user is interested in the relation of spatial properties of a target and its context (whether it is in the physical or thematic layers). Assessing this particular query is especially difficult, given the lack of analysis details in some papers. Therefore, for this classification, we looked for the use of keywords such as "context", "neighborhood", and "vicinity" in the description of the requirements, methodology, or analysis. A large number of surveyed papers (129) could then be classified as performing this query. The topology-based system proposed by Doraiswamy et al., for instance, highlights the need for *3D context models* and *explore the view extent over the city* [DFL*15], while the study by Moghadam et al. assesses the concentration of energy consumption over neighborhoods [MTML18]. Spatial relation is also important in wind analyses [EGAB17, AKH19], given the impact of neighboring buildings on the aerodynamics of a tall building.

7.3. Analysis targets

Lastly, we classified papers according to their primary abstract targets. We included five targets covered by Munzner [Mun15]: trends, features, extremes, distributions, and outliers. The target of analysis was considered with respect to the thematic layer. A trend target is a pattern in the data, e.g., the high energy consumption of apartment units during summer. For example, Nouvel et al. compare the impact of data quality of 3D buildings on urban heating demand modeling, analyzing trends over different city districts [NZCE17]. A feature target is any structure in the data that might be of interest in the analysis task, e.g., summer energy consumption within certain blocks of a city. Chen et al. propose an exploded view technique for immersive urban analytics in which the feature target is the road network occluded by buildings [CQW17].

An extreme target is the minimum or maximum value when considering a specific range, e.g., maximum summer energy consumption for apartments in a building. Cornel et al. propose an uncertainty-aware visualization of vulnerability that includes extreme values across flooding scenarios [CKS*15]. Hu and Wendel evaluate minimum and maximum values in their anisotropy analyses over two neighborhoods in New York City, leveraging 3D building data [HW19a]. Delmas et al. optimize building placement to maximize solar potential [DLD*16], and De Luca and Voll analyze daily minimum sunlight hours [DLV17]. View impact analyses such as the paper by Yu et al. are also interested in extreme targets to identify floors with the least occluded views of landmarks [YYS*16]. A distribution target is one where the user is interested in the spread of values considering a range, e.g., distribution of energy consumption for apartments in a building. The papers classified as systems usually support both exploration and summarization, therefore enabling the visualization of spatial [LXWY22] and spatiotemporal [MDL*19, MFS*23] distributions. Finally, an outlier target is the one that deviates from that trend, e.g., a specific apartment with low energy consumption during hot summer days. Even though no paper explicitly mentions outlier as their analysis target, the target is supported by many of the visual encodings used by the surveyed papers, as we detail in Section 10.1.

8. Data dimensions (What)

The 3D datasets used for the different use cases covered in this survey impose several challenges to all stages of the urban data life cycle. From modeling 3D geometries and deploying physical sensors to developing computational strategies capable of sampling inherently 3D urban-related phenomena and aspects, the collection/generation and management of 3D data is either expensive, computationally intensive, or both. Also, 3D data is provided in a myriad of different formats and resolutions depending on their sources, type of information, and analysis requirements. For this

reason, curating, managing, analyzing, and visualizing these data require specifically designed data solutions.

In this section, we will then highlight What datasets and properties were used in the surveyed works. Given the aforementioned challenges and the tight connection between visual analytics and data, in this survey, we identified four data dimensions to categorize the reviewed papers: to highlight the connection between physical and thematic layers, we have included a category with the physical entities considered in the data – i.e., buildings, streets, nature (Section 8.1); the source of the domain-specific 3D thematic data (Section 8.2); the properties of such data where we cover whether the data is uniform, multivariate, volumetric, temporal, and has an attached structural meaning (Section 8.3); and the urban spatial coverage of the data – i.e., micro, meso, macro (Section 8.4). Figure 7 shows the distribution of tags according to the What and How dimensions.

8.1. Physical data entities

In conjunction with thematic data, 3D urban data papers also utilize data describing the built environment – called here physical data. Physical data is fundamental since it gives geometric support and spatial context to the thematic data. Depending on the resolution of both the physical and thematic data, the latter must be aggregated over the former. In our survey, we have considered three types of physical data entities: buildings, streets, and nature, with nature encompassing parks, trees, mountains, hills, and waterfronts. Figure 8 shows an example of visual analytics tools making use of multiple physical layers and aggregating shadow contributions at a building level. The topography of the urban environment is particularly important for papers tackling disaster & damage management. Cornel et al. [CKS*15, CBK*19], Vuckovic et al. [VSOC21], and Waser et al. [WKS*14] all make use of elevation data for both modeling and visualization purposes.

In the surveyed works, OpenStreetMap [HW08] is the primary source of data for buildings, streets, and certain nature-related features, such as parks and waterfronts. OpenStreetMap makes data available as a series of crowdsourced tags attached to data structures representing nodes, ways, and relations. Buildings, for example, are defined as 2D outlines with associated heights (or number of floors) and roof types, requiring adequate data processing to extract polygon meshes for visualization. Alternatively, authoritative sources usually make building data available using the CityGML standard with geometry information specified at a certain level of detail, or directly as polygon meshes. Government agencies are the primary source for tree data, usually made available as tabular data with the location of trees. For topographic features, in our survey, we have found that works usually make use of terrain elevation rasters made available by OpenTopography and/or NASA's Shuttle Radar Topography Mission. Point cloud and image data are also potential sources for data describing the built environment, though they require advanced techniques to generate polygon meshes and attributes to be leveraged by visual analytics tools [BI21, HYK*21, HYK*22].



Figure 7: Distribution of surveyed papers according to What and How dimensions, with shades denoting tag occurrence.

8.2. Thematic data origin

The thematic data considered in the surveyed works were classified according to their origin, i.e., how they were acquired or generated. Given that many of the considered use cases depend on modeling and simulation approaches, such as view impact, sunlight access, wind, and seismic analysis, it is no surprise that most works relied on simulation-based datasets. In this classification, we have also included data created from mathematical models, such as CFD [AC17, LMBZ18, NSN18, AC18, KKK21, LMS*23], WRF [WHN*21, KPMB21, HYF*23], and ENVI-met [HHG23].

The second most popular data origin was data acquired using sensing methods. In this category, we include remote sensing [LD10, CRPB14, BK19], data gathered by robots [BSK*17] or by sensors deployed in the urban environment [OS18, MKG21]. Palliwal et al., for example, leverage sensors placed along the façade of a building to estimate urban farming potential [PSTB21].

The third data origin category in our work covers data that come from authoritative sources and/or surveyed data, such as land use description [FLD*15, KD17, ZY18], demographic data [CWK*07], public infrastructure information [BG19], or property cadastre [SAL*20]. We have found that survey data is usually used as a criterion to filter the data in certain visualizations. For example, both Urbane and VitalVizor have land use attributes as dimensions in their brushable parallel coordinates.

Finally, we have also considered approaches that derived new data from raw data using computational models, including machine learning ones. Examples include approaches for urban energy modeling [AAA*19], wind flow [MSD20, KCG22], and view impact [LXWY22]. It is interesting to point out that we observed a growing number of learning-based 3D data generation approaches in the last five years, which is expected given the omnipresence of neural network-based solutions nowadays.

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd.

8.3. Thematic data properties

In our survey, we found that thematic data has distinct spatiotemporal characteristics. Regarding their spatial information, 3D thematic datasets can be categorized into two broad categories: (1) data defined at the surface of physical data entities (e.g., buildings) and (2) data defined inside 3D volumetric regions, not necessarily inside buildings. We can further subdivide category (1) into two subcategories: data uniformly discretized on the surface of physical data entities (e.g., solar radiation exposure on different parts of the building), not necessarily aligning with real-world structures like windows and floors (Figure 4 (data creation)), and data discretized into smaller units based on physical aspects with semantic meaning, e.g., structural elements such as apartment units, windows, floors, massings (Figure 4 (System)). In our classification, we use uniform and structural to refer to these two categories, respectively. The choice between uniform and structural discretization is based on the use case of the paper. For example, energy modeling [ABLD*22] and 3D property cadastre [SKB*13] associate data with individual buildings, floors, or



Figure 8: Examples of integration between physical and thematic layers for sunlight access analysis (left, [MDL*19]) and disaster mgmt. (right, [CBK*19]).

property units. Therefore a structural discretization is more appropriate. Out of the eight property cadastre surveyed works, four make use of structural discretization. On the other hand, seismic simulations [XLH*15] primarily rely on a uniform discretization along the building's surface. Some other use cases, such as sunlight access and view impact analysis, make use of both uniform (e.g., view impact computed on the surface of the building) [DFL*15, MDL*19] and structural discretizations (e.g., view impact computed for each building unit) [FLD*15, TCFR21]. In category (2), data defined inside 3D volumetric regions, we have data produced by simulations and/or related to natural phenomena (Figure 4 (Application studies)), such as flood and heavy rain [CBK*19], wind [HP20, SDP20, ZKL*21] and sound or radio propagation [BCC*18, TLH*22]. Such data is particularly popular in papers leveraging CFD wind data, with 74% making use of volumetric data.

We also classified papers on whether they used temporal data. Since urban regions are dynamic and affected by human behaviors, natural phenomena, and other elements that are strongly associated with periods of the day, days of the week, and seasons of the year, it was no surprise to discover that many of the selected papers consider thematic datasets containing a temporal component. For example, Mota et al. present an evaluation of different designs for visual analytics tasks considering sunlight data for the different seasons of the year [MFS*23].

Finally, we classified the thematic data based on whether they make use of multivariate data. View impact analysis is an example of a use case that takes into account several attributes to compute visibility scores (e.g., view scores to greenery and land-marks). Turan et al., for instance, take into account four different attributes in their view impact analysis [TCFR21]. Visualization systems [FLD*15,OSS*17,MDL*19,ZZL21] usually support multidimensional datasets through standard or new visualization encodings. For example, Waser et al. propose a novel interface for the visualization of multidimensional ensembles targeting the problem of flood management [WKS*14].

8.4. Spatial data scopes

The surveyed papers focused on three different spatial resolutions for their analytical tasks. We have identified approaches that consider specific sites (e.g., façade [PSTB21] or buildings [PTD13, DLV17]), multiple sites (e.g., one or several blocks [Zha13, MnPBM*17, KKO18, KKK21, LXWY22, ZWK*22]), or large regions (several neighborhoods or the entire city [FLD*15, MDL*19, CMF*22]). We call these approaches micro, meso, or macro scale, respectively. Figure 9 (top) presents three examples of analyses performed at different scales: view impact using Urbane at the macro scale [FLD*15], seismic analysis by Xi et al. at the meso scale [XLH*15], and sunlight access analysis by Wu et al. at the micro scale [WHG10].

It is interesting to notice that analysis across multiple scales is performed in a number of domain papers. Figure 9 (middle, bottom) shows a view impact analysis performed across the three different scales [LXWY22, YYS*16]. In these analyses, the domain expert is not only interested in an individual building but also in the



Figure 9: Examples of different analysis scopes. Top left: view impact analysis at a city scale [FLD*15]. Top center: seismic analysis at a neighborhood scale [XLH*15]. Top right: shadow impact analysis at a building scale [WHG10]. Middle and bottom: two domain papers with view impact analysis across the three scales [LXWY22, YYS*16].

interaction between multiple neighboring buildings and the spatial distribution over different neighborhoods.

9. Task characterization

Designing visualizations for 3D urban data involves the careful consideration of data and intended analyses. In this section, we use the distribution of What and Why tags to discuss relevant tasks across the different use cases considered in this survey (Figure 10). We begin with six guiding questions (Q1-Q6). Questions are on what are the most common search actions, query actions, analysis targets, thematic data properties, spatial scopes, and physical data entities. We then use the distribution of tags along use cases, i.e. the answers to Q1-Q6, to derive common visualization tasks. This methodology follows the approach of Mota et al. [MFS*23], but using the distribution of tags within a use case to infer its common ness.

9.1. Guiding questions

To answer questions Q1-Q6 about the occurrence of tags, we consider that a specific tag is common with respect to a use case if it is associated with more than 50% of the surveyed papers of that case. At the end of each question, we summarize the most common use cases for each tag.

Q1. What are the most common search actions?

The majority of the surveyed studies analyze data at a specific location but without knowing the identity of the target – i.e., a browse (0 + 44) search action. In our survey, we have found that such action is common in all considered use cases. However, we found that a locate ($\ +44$) search action is only common in disaster management (77% of the papers) and noise (50%) use cases. Similarly, an explore ($\ +44$) action is also common in disaster management (62%) and noise (50%) use cases. In our review, we have identified fewer than ten works that make use of lookup search ($\ +44$).

Browse: Common in all use cases; Locate: Disaster mgmt. and noise; Explore: Disaster mgmt. and noise.

Use case Sunlight access Wind View impact Energy ēŌ ŌŌŌŌŌ ā Ē ōē 00000000 $\otimes \otimes$ $\otimes \otimes \otimes$ -1 Disaster mamt. ĀĀĀĀ 0000 iño Urban climate ÓÕÕÕ ř OOOOO 000 Noise 00000000 -Property cadastre ōō ŏ _____ Others Buildings DOOOOOOO 00000000 00000 Nature 000000000 $\cap \cap \cap$ 00000 Uniform Structural 0000 00000 Multivariate ŌŌ ē ŌŌ Volumetrio Temporal Sensing 00000000 000 Simulation Derived Surveyed 00000000 00000000 00000 000 Micro ēēēē ē 00000000 ŏŏ WHV

Figure 10: Distribution of surveyed papers according to What and Why dimensions, with shades denoting tag occurrence.

Q2. What are the most common query actions?

We have found that identity and spatial relation actions are common in all use cases. For example, over 85% of the surveyed papers tackling disaster management consider both identify and spatial relation tasks. Among wind studies, all of the papers make use of spatial relation actions, given the fundamental role that the built environment plays in determining wind patterns. The summarize action is common in four use cases: noise (63%), energy (52%), disaster management (54%), and view impact analysis (50%). The compare action is common in all but the property cadastre case (only 25% of the papers).

Identify: Common in all use cases; Spatial relation: Common in all use cases; Compare: Common in all except property mgmt; Summarize: View impact, energy, disaster mgmt., noise.

Q3. What are the most common analysis targets?

The analysis of features is common among all of the surveyed use cases, with more than 50% of the papers in each category considering it. The analysis of distributions is popular in all of the use cases, with the exception of property cadastre. The analysis of extreme values is only common in noise studies (63% of the papers in that category), and we found that analysis of trends and outliers is not common in any use case.

Features: Common in all use cases; Distribution: Common in all except property cadastre; Extremes: Noise.

Q4. What are the most common thematic data properties?

The presence of data uniformly discretized on the surface of physical objects is common in four different use cases: urban climate (90%), noise (88%), wind (88%), and sunlight access (59%)

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd. analysis. Conversely, the use of data discretized at a structural level (e.g., floors) is only common in the property cadastre case, where data is associated with individual floors or property units. In our survey, we have also found that multivariate data is common among urban climate (77%), disaster management (73%), and view impact (50%) cases. Temporal data is common in disaster management (85%) and urban climate (53%), and volumetric data is common in wind (74%) and urban climate (53%) cases.

Uniform: Urban climate, noise, wind, sunlight access; Structural: Property cadastre; Multivariate: Urban climate, disaster mgmt, view impact; Temporal: Disaster mgmt, urban climate; Volumetric: Wind, urban climate.

Q5. What are the primary spatial scopes?

We have found that micro-scale analyses are common in all of the surveyed use cases, particularly sunlight access (87%), wind (81%), and disaster management (77%). In the surveyed papers, mesoscale analyses are popular in all use cases, with the exception of wind and property cadastre. Macro-scale analyses, on the other hand, are common in disaster management (54%).

Micro: Common in all use cases; Meso: Common in all except property cadastre; Macro: Disaster mgmt.

Q6. What are the primary physical data entities?

As expected, given the scope of the survey, building data is a common data entity across all use cases. Street data is commonly leveraged in use cases studying urban climate (100%), wind (88%), noise (75%), and disaster management (69%). On the other hand, nature data is not common in any use cases, being moderately popular only in disaster management (38%) and view impact (29%) cases.

Buildings: Common in all use cases; Streets: Urban climate, wind, noise, disaster mgmt.

9.2. Common analysis tasks

Common visualization tasks can be derived from the combination of the tags and their frequency that we derived in Q1-6 in the previous section. Next, we highlight six of these common tasks across multiple use cases (**T1-6**). While not exhaustive, these tasks are diverse in their analytical (Why) and data (What) dimensions. We also note that these tasks are oftentimes combined, e.g., analysis of volumetric data (T5) can be accompanied by uncertainty analysis (T6). In subsequent sections, we highlight how different visualization techniques facilitate these tasks.

T1. Micro-scale browsing for site identification and comparison

A common task in urban use cases is the building or façadelevel analysis for the identification of potential sites matching certain requirements for equipment installation (e.g., solar panels) or urban redevelopment. For example, urban planners are interested in the analysis of the distribution of sunlight access data where the general location is known (a single building $\mathbf{0}$) but not the specific range of values **bu** (browse). The goal is to identify suitable locations in façades for urban farming or for green visibility. In this micro-scale analysis, data is uniformly discretized at a building's façade (see Figure 9 (middle, right and bottom,

right)) [PSTB21, LXWY22, YYS*16]. Also, architects interested in the pedestrian level of comfort leverage wind and sunlight access data at the building and street layers. The analysis involves the comparison of wind comfort maps across multiple months [DL19].

T2. Micro-scale browsing for individual unit identification

In the property cadastre case, the target data is a volume composed of individual units (e.g., apartments, offices). In other words, individual units assemble into a dense 3D building volume. While in **T1**, experts were only interested in the outer regions of a building, in property cadastre tasks, architects, land owners, and city officials need to assess attributes from inner units that are potentially occluded. These attributes include primary function, ownership information, and legal boundaries. After an individual unit is identified, an expert can further perform shadow or visibility analyses [SKB*13], or study the 3D morphology and spatial positions of a building [YCL*19].

T3. Multi-scale exploration for building site identification

It is common for architects and urban planners to rely on analyses over multiple scales to help identify sites for the development of new buildings. Differently from **T1**, which is restricted to a known building, here we have an exploration where both the location and identity are unknown. At each level, the expert will compare distributions leveraging different data. At a macro scale, an expert usually uses spatial data to compare neighborhoods and better understand strengths and weaknesses between them to establish performance thresholds [FLD*15]. At a meso scale, an expert establishes criteria to assess the consequences of new buildings at a neighborhood level, which can include multivariate attributes to measure view impact [OSS*17]. At a micro scale, they can then study the impact of building massing configurations on urban density [ABLD*22] and sky exposure at street level.

T4. Multi-scale analysis for scenario comparison

After identifying a particular site (such as discussed in T3), it is common for experts to have to compare different scenarios for what-if analyses. In disaster management, for example, these can be floodwall breach scenarios [RWF*13, CKS*15] with data defined at building (e.g., the height of flood) and street (e.g., flooded areas) levels. In building energy modeling, the scenarios are different building type parameters (e.g., single-family homes, dense high-rises) that are considered to target certain carbon reduction or energy efficiency goals [ABLD*22]. In view impact and sun access analyses, a scenario is usually a building massing configuration (e.g., number of floors, floor plate size) that impacts the surrounding context [FLD*15, DFL*15, OSS*17, MDL*19]. The task then involves analyzing the trade-offs and understanding the potential impact of these interventions (floodwall breach or new buildings) at both micro and meso scales. Scenario comparison and analyses across scales are also increasingly more common in the urban climate use case [PRd19, JN20, JK21], using multi-scale models that couple micro (e.g., numerical models such as CFD) and meso and macro-scale approaches (e.g., WRF, remote sensing maps).

T5. Micro and mesoscale analysis of volumetric data

To better understand the impact of wind on new developments and neighborhoods, experts are increasingly turning to computational fluid dynamic or ENVI-met simulations. These simulations take into account 3D models and dominant wind directions and generate a 2D (at street level) or 3D time-varying flow volume with wind speeds and/or temperature. The primary task of the expert is to then visualize distributions and (1) identify trends (e.g., increase in wind pressure) at both micro (i.e., individual building facade [AKH19]) or meso (i.e., a neighborhood [HP21]) scales, (2) values above certain thresholds [HYF*23], and (3) individual features such as jet patterns [KBHS18], channeling, sheltering or wake effect [EGAB17].

T6. Analysis of uncertainties

We have also found that the analysis of uncertainties is common across different use cases, particularly the ones leveraging data from simulations or models. For example, when considering flooding simulations, multiple scenarios need to be considered and evaluated for effective disaster prevention [RWF*13]. In noise mapping, given the complex and dynamic nature of the urban environment, predictions can suffer from underestimation [LJR*14]. The primary task of an expert is to then explore this uncertain data at the macro (e.g., to identify the noisiest areas at building facades or streets) or meso (e.g., to identify effective interventions during disaster events) scales.

10. Visualization & interaction dimensions (How)

A fundamental goal of urban visualization is to allow users to investigate the physical properties of a city and relate them to thematic data and vice versa. In the case of 3D urban visualization, these physical properties are 3D representations of buildings, streets, or trees, and thematic data, that usually describes complex threedimensional phenomena, such as wind, microclimate, shadow, visibility, or flooding.

The design of an effective 3D urban visualization depends on how the thematic data is encoded (Section 10.1) and by which means they are visually integrated with respective physical data (Section 10.2). Figure 7 summarizes the distribution of How and What tags. The extent and verticality of modern cities lead to common challenges such as occlusion and higher cognitive load associated with frequent viewpoint changes. To enable users to perform the tasks described in Section 7, an effective design needs to deal with occlusions (Section 10.3) and support users in navigating urban space (Section 10.4). Further, we look at how visualization and computational methods are integrated (Section 10.5) and which display modalities are used (Section 10.6). Then we conclude this section on how authors typically evaluate their respective contributions (Section 10.7). In this section, we will also highlight How visualization is used to execute the previously identified tasks.

10.1. Visual encodings

The integration of physical and thematic data visualizations and the way thematic dimensions are encoded are closely related. We chose to first give an overview of the encodings we have encountered and then support them with examples to facilitate the discussion of their interplay in Section 10.2. In the following two sections, we distinguish between spatial encodings where data is visualized directly in the 3D urban scene, and non-spatial encodings where data is visualized in the form of 2D graphs.

Miranda et al. / The State of the Art in Visual Analytics for 3D Urban Data



Figure 11: Different approaches to visualize different urban data defined inside 3D volumetric regions. (a) Shadow volumes [FWC22]. (b) Realistic water rendering [CBK^{*}19]. (c) Line integral convolution and 3D streamlines [KBHS18]. (d) Arrow glyphs [ZKL^{*}21].

10.1.1. Spatial encodings

We distinguish two cases spatially encoding data in a 3D urban visualization. The first is mapping, where thematic data has been mapped and visualized onto physical data and the second is 3D representations to visualize the volumetric context of urban environments.

Mapping onto surfaces. When mapping data onto buildings, we have found different granularities related to the properties of thematic data discussed in Section 8.3. Nouvel et al. color entire buildings using a discretized color ramp to illustrate their heating demand [NZCE17]. Mapping color in this granularity is also typical for encoding flooding damage to buildings on a neighborhood scale [WKS*14, RWF*13, CBK*19, YNK*18]. In the work of Vanegas et al., buildings are colored depending on their sunlight exposure, the distance to the next park, or their floor-toarea ratio [VGDA*12]. Tang et al. present noise-mapping data discretized and visualized per floor level (i.e., discretized over structural units) [TLH*22]. Wolosiuk and Mahdavi [WM20] and Li et al. [LXWY22] use a uniform fine-grained mapping to show detailed incident solar radiation on a roof and office view quality on the façade of high-rise buildings (Figure 9 (bottom,right)), respectively. The work by Lorenz et al. [LD10] discusses a graphics approach to map 2D rasters and 3D geometries. Such visualizations directly support the micro-scale façade-level analyses in T1.

Phenomena that affect both buildings and street and ground levels can be encoded using a 3D map that covers all physical data, for instance, for light [MnPBM*17], shade [CFL14, MDL*19], wind [AKH19], noise [ZLH*17], and radio propagation [BCC*18]. Thematic data encoded in 2D map layers can also be combined with the 3D urban model, for instance, showing the density of subway stations [FLD*15] or wind force at pedestrian level [YNK*18]. However, in these cases, we have found that a fine-grained color mapping onto building façades is omitted. Given the nature of the data considered in this survey, it is no surprise that 3D maps are common across all use cases. This is largely due to the inherent capacity of 3D visualizations to provide a richer representation and better contextualization of spatial relationships and depth than their 2D counterparts. 2D maps are frequently utilized in wind and urban climate studies to display data slices at pedestrian height (~1.75 meter). These maps help visualize and analyze information related to airflow patterns, temperature distribution, and other environmental factors crucial in these cases.

3D representations. Phenomena that are defined inside 3D volumetric regions (see Section 8.3), for instance, shade, water, or

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd.

wind, are often encoded into 3D volumetric representations. Fijałkowska et al. show shadow volumes to illustrate when and how many photovoltaic panels are in the shade [FWC22]. Cornel et al. present a technique to reconstruct a water surface from large-scale shallow water simulation [CBK*19]. They also enhance the visualization of the water by waves and foam to indicate flow direction and barrier overtopping, respectively. Beran et al. [BJK*22] propose a set of methods for space-time cube visualization of noise data. Analysis of wind effects is typically preceded by a CFD simulation. There is a large design space visualizing such data [FH09], but most works choose to use glyphs / streamlines in this context and use the color channel to encode wind speed [WBH14,EGAB17,AKH19,HP20,ZKL*21,KKK21,Ath22] or temperature [HP21]. Koch et al. use illuminated streamlines accompanied by a 2D map using line integral convolution to show wind force on the pedestrian level [KBHS18]. Another visualization pairs 3D streamlines showing the wind force with a 2D map encoding the ground temperature, similar to Allegrini and Carmeliet [AC18]. Zhang et al. [ZKL*21], and Hadavi and Pasdarshahri [HP21] additionally use arrow glyphs on slices of colorcoded 2D maps to indicate wind direction. Figure 11 highlights some of the approaches.

It has to be noted that despite the fact that the visualization of CFD data is a core competence of the scientific visualization community, we have not found any systems or design studies dealing with wind analysis in an urban context. Gaultier et al. use three different geometric representations – vertical planes, horizontal planes, and points – to visualize temperature data together with building geometry [GBC20]. To visualize the seismic response of building sin an earthquake scenario, Xiong et al. deform the 3D building geometry and color-code the displacement [XLH*15]. Chen et al. place a ladder truck in the urban scene and illustrate how specific windows can be reached supporting a fire-fighting scenario [CWS*17]. These visualizations support the **T3** previously identified.

10.1.2. Non-spatial encodings

We have found a diverse set of non-spatial (i.e., abstract [SOP*15]) visualizations in the surveyed articles. We also included static visualizations produced in papers that were classified under the application study type. These are not interactive and rarely integrate with spatial views, though we believe they can inform effective visual designs as domain experts are already familiar with these encodings. Scatter plots, bar charts, and line graphs are widespread among visualization and domain articles across all use cases. For example, Miranda et al. use linegraphs to illustrate changes



Figure 12: *Examples of approaches to integrate physical and thematic data: superimposition* [EGAB17], *embedded views* [CKS*15], *linked views* [ZY18], *interchangeable* [KK19], *and juxtaposition* [ZKL*21].

in light and shadow scores over the course of a year [MDL*19]. Ribicic et al. visualize flood simulation outputs in bar chart histograms and scatterplots to filter interesting runs [RWF*13]. Krietemeyer et al. encode "occupancy schedule," "equipment energy use," and "average energy use of a workday" into two bar charts and a line chart, respectively, as details-on-demand for a building [KK19]. In the case of wind analysis, we observed frequent use of stacked radial histograms, also referred to as rose diagrams, encoding the frequency of wind of a certain strength and direction [GBI14, SDP20, AKH19]. Fijałkowska et al. employ rose diagrams to show the daily shade and wattage of solar panels [FWC22]. Hu et al. use polar plots where a 2D scalar field is plotted with respect to zenith and azimuth angles, basically describing a near-hemisphere [HW19a]. Heatmaps are employed to display temporal patterns [OS18, HW19a], but their axes may also encode positions on a building façade to show good spots for urban farming [PSTB21], radiation variation [CRPB14] or show which windows have the best views [LXWY22].

When it comes to encoding multiple thematic dimensions, the **parallel coordinates** plot (PCP) is a popular choice in the visualization community. The PCP can show the plethora of thematic values of a building with relation to all other buildings [FLD*15, ZY18] or neighborhoods [BDW*08, CWK*07] and allows users to assess the performance of building candidates [ZZL21] and designs [DFL*15]. PCPs are also used in disaster management to compare the performance of barriers [RWF*13] and rescue paths [BSK*17]. PCPs are usually employed in multiscale analyses (**T3** and **T4**) as a way to filter geographical areas that satisfy certain constraints.

There is no PCP in domain contributions except the work of Wilson et al. [WDDH19] and Su et al. [SD23], where they combine design parameters and urban indicators of procedurally generated block layouts. This might be owed to the unfamiliarity by domain experts but also due to the fact that, in order to be effective, a PCP needs to support interactivity, such as linking & brushing. Analysis articles rather use a scatterplot matrix (SPLOMs) or a correlation matrix to show multiple dimensions at once, for instance, to illustrate correlations between building characteristics and performance indicators [ZLH*17, MTML18, BK19]. Ortner et al. use an interactive matrix of candidate buildings and viewpoints where each cell encodes four different view impact metrics as bar charts [OSS*17]. Bock et al. use an interactive SPLOM in addition to their PCP [BSK*17]. In order to support T6 and uncertainty

analyses, domain papers usually employ well-known visualization methods in a static fashion, such as box plots [LJR*14].

10.2. Integration of physical & thematic visualizations

To structure this section, we expand on the related work categorization presented by Mota et al. [MFS*23] and derive the following ways of visually integrating physical and thematic data: superimposition, embedded views, linked views, interchangeable, and juxtaposition. Figure 12 shows examples of works using these approaches.

Superimposition displays two or more data instances in the same coordinate space. This comes down to the 2D or 3D mapping of thematic data onto physical entities as described in the previous section. Also, the integration of 3D streamlines, noise, or sunlight access data falls into this category (T5), which is the dominant way of combining thematic and physical urban aspects in the surveyed literature. A variation of this concept involves streamlining slices with different heights [ZKL*21] and orientations [EGAB17] (Figure 12 (Superimposition)). Hadavi and Pasdarshahri [HP21] color-map wind speeds on a semi-transparent vertical plane in addition to a colored 2D layer with arrow glyphs at pedestrian level. For noise and sunlight access data visualization, the majority of works overlay the data on the surface of buildings [RGN12, MDL*19, LX21, WUAC23]; a notable exception is Beran et al. that propose the use of spacetime cube isopleths [BJK*22].

In embedded views, views establish their own coordinate system but are spatially anchored in the physical domain [MFS*23]. These views combine 2D and 3D graphical elements, in the form of charts and 3D maps [DRST14]. For example, Cornel et al. use circular glyphs embedded on the 3D environment to display flooding uncertainty [CKS*15] as shown in Figure 12 (Embedded view). Butkiewicz et al. anchor histograms and PCPs with a probe metaphor to neighborhoods displaying respective demographic data [BDW^{*}08]. A similar approach is used by Vuckovic et al. to place small hydrographs [VSOC21] (Figure 13 (left)) and by Wilson et al. to anchor performance indicators in the form of bar charts to urban layouts [WDDH19]. Ortner et al. use color-coded circular glyphs to mark viewpoints in the scene that are impacted by candidate buildings [OSS*17]. Mota et al. evaluate the efficacy of visualizations that are embedded within the physical domain, such as the bar charts shown in Figure 13 (right). In all cases, embedded views show detailed thematic information that is confined to a cer-

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd.



Figure 13: Examples of embedded views used at two distinct scales. Left: Vuckovic et al. [VSOC21] showing a wider spatial extent. Right: Mota et al. [MFS^{*}23] depicting a local spatial context.

tain region, block, or building. In our survey, visualization papers were the only examples of the use of embedded views.

Orthogonal to the concept of embedding views is the coordination and linking of views (Figure 12 (Linked view)). For instance, selecting a water line in the façade area plot loads the respective flooding scenario that led to that line [CKS*15]. However, to avoid clutter and occlusion, non-spatial visualizations are often juxtaposed with the spatial view. Then linked highlighting or filtering is a way to anchor the encoded thematic data in the physical domain and vice versa. A majority of the interactive systems from the visualization community allow linked interaction to highlight and/or filter data. In contrast to embedded views, non-spatial views in the reviewed cases typically display an overview of all entities instead of a regional subset. Examples are all solutions featuring a PCP [RWF*13, FLD*15, DFL*15, BSK*17, ZY18, ZZL21]. In the case of Krietemeyer et al., the non-spatial views display detailed information about a single building, and the linked highlighting is realized by brushing the temporal domain [KK19]. Linked views have been extensively used to support multi-scale analyses (T3 and T4), given that different scales might require the combination of multiple types of encodings.

For interchangeable integration, the same view is used to show different states requiring a dynamic visualization. In Vis-A-Ware, users can flip through building candidates or show them all at once (Figure 14 (Ghosting)) [OSS*17]. Krietemayer et al. let users browse through different time steps via a filmstrip metaphor (Figure 12 (Interchangeable)) and also provide a slider-based interaction to illustrate before and after scenarios [KK19]. With the Shadow Profiler, users can step through or animate the course of shadows cast on an urban scene [MDL*19].

Authors often juxtapose spatial views when different states of the physical domain shall be compared. This works especially well when the spatial domain can be reduced to a region of interest such as a city block [WDDH19] or a single building [DFL*15]. In the domain of urban climate and wind analysis, we have found several application studies that convey their results by juxtaposing static visualizations. Wilkinson et al. illustrate the difference between a ground truth simulation and their approximate wind interference model by two top-down views showing streamlines and three views of a color-mapped building, where the third view encodes the difference [WBH14]. Elshaer et al. extensively use this concept, most notably circular cut-outs that show a vertical and a



Multi-view

Assisted

Figure 14: Examples of approaches to handle occlusion: Deformation [CMF^{*}22], ghosting [OSS^{*}17], slicing [KPMB21], bird's view [WSK18], multi-view [PTD13], assisted-steering [ZZL21].

horizontal streamline slice for three different time steps [EGAB17], similar to Zhang et al. who show circular urban cutouts overlaid with horizontal streamline slices at different heights (Figure 12 (Juxtaposition)) [ZKL*21]. Hadavi and Pasdarshahri juxtapose four views showing two different buildings overlaid with detailed 3D streamlines with color-coded velocity in the upper row and color-coded temperature below [HP21]. Pasewaldt et al. juxtapose the 3D view (perspective view) of a building and a 2D view (multi-perspective view) of its unwrapped façade, both mapped with thematic data from solar potential analysis [PTD13]. Juxtaposed spatial views have been found to be particularly popular in papers where the primary task is the comparison of multiple scenarios (**T4**) [DFL*15,OSS*17].

In our review, we have found that ArcGIS [JVHKL01], Urban-Sim [NBW03] and QGIS [qgi] support superimposition, linked view, interchangeable and juxtaposition. Kepler.gl [kep] and deck.gl [Wan17] support superimposition, linked view, and interchangeability but have limited support for juxtaposition. None of these tools and libraries provide support for the creation of embedded views in the 3D environment.

10.3. Occlusion handling

The verticality and density of cities easily lead to the occlusion of analysis targets. As a result, the analysis tasks described in Section 7 are impeded or may fail. We classify the occlusion handling measures of the surveyed articles based on six categories: deformation, ghosting, multi-view, assisted steering, slicing, and bird's view. This is an adaptation of the five occlusion handling design patterns introduced by Elmqvist and Tsigas [ET08]. Most notably, we changed "tour planner" to assisted steering, which conveys the more interactive nature of the found techniques, and we added bird's view, which is a ubiquitous remedy for occlusion problems in an urban context. We show examples for each technique in Figure 14.

Several deformation techniques alter the 3D urban model primarily by scaling to disocclude routes and landmarks [HWC*09, HWAT13]. In addition to disocclusion through scaling, Deng et al. also guide users to a bird's eye perspective and demonstrate the potential of their system by encoding a 3D pollution layer [DZMQ16]. Wu and Popescu render multiple view angles from a single position and compose them to a distorted view of an urban scene [WP16]. Chen et al. present a method for dynamically "exploding" urban geometry to reveal a point of interest on the ground in an AR setting [CQW17]. Ying et al. propose a similar approach for the identification of individual units in property cadastre (T2) [YCL*19]. To increase viewing distance in a CAVE setup Engel et al. bend a sparse city scene upwards [EPTD12]. In a fully immersive VR setting, Chen et al. wrap the dense cityscape of Manhattan upwards onto a cylindrical projection to keep distant landmarks in sight (Figure 14 (Deformation)) [CQW17].

Cornel et al. let users focus on a target building and cut away all occluding geometry while using ghosting for the outlines of adjacent buildings for context [CKS*15]. The system presented by Lv et al. enables users to inspect individual floor layouts. When a floor is selected, the floors above it are cut away [LLZ*16]. Seipel et al. investigate how different transparency settings influence a person's perception of complex 3D real property cadastre data [SAL*20]. Gaultier et al. use ghosting on thematic data encoded by 3D glyphs to blend temperatures with the urban geometry [GBC20], supporting **T5**. Robineau et al. adjust the transparency of built environment features to highlight wind streamlines [RRM*22].

Pasewaldt et al. take a multi-view approach where the 3D view of a building is supported by a multi-perspective view that unwraps the façade onto a 2D plane effectively removing occlusion (Figure 14 (Multi-view)) [PTD13]. A similar approach has been employed to visualize the distribution of view impact [SMS23], temperature [HHG23], and sunlight access [CRPB14, PSTB21] over building façades. Zhang et al. support user navigation with assisted-steering by automatically computing optimal viewpoints which try to minimize occlusions (Figure 14 (Assisted)) [ZZL21]. Deng et al. [DZMQ16] and Li et al. [LLC*20] were the only other works that made use of assisted steering.

Another popular approach, particularly for volumetric data in wind, urban climate, and noise use cases, is the use of slice planes. The concept is similar to volumetric probes, in which the occlusion of targets is handled through the use of user-defined cutting planes (Figure 14 (slicing)). In the surveyed works, we have found two common slice-based approaches. In the first group, we have works that make use of *static* horizontal and vertical cross sections [SAYJ18, JN20, SAH20]. For example, Shaeri et al. use vertical cross sections at different sections of a neighborhood and horizontal cross sections at different heights for the analysis of air speed and air pressure [SAYJ18]. Another approach is to use cross-sections embedded in an interactive 3D environment. For example, Kadaverugu et al. superimposes vertical cross-sections within a 3D urban environment to slice a CFD domain [KPMB21], and Ranjbar et al. and Beran et al. use cross-sections for 3D noise visualization [RGN12, BJK*22].

Occlusion handling is a common topic in 3D visualization research [ET08], and several contributions exist in the context of 3D urban scenes. However, they barely find their way into 3D urban analytics solutions that feature rich thematic data, as they are present in most use cases. The majority of spatial visualizations rely on the users taking a bird's view through manual camera manipulation. Naturally, this is the preferred method for noninteractive visualizations, such as in Willenborg et al. (Figure 14 (Bird's view)) [WSK18] and most other application studies.

10.4. Navigation methods

Navigation poses one of the main challenges when visualizing 3D urban environments. Observing physical and thematic data at different scales and heights impeded by occlusion requires frequent viewpoint changes that lead to a loss of context and can put a high cognitive strain on the users. In accordance with the travel metaphors presented by LaViola et al. [LJKM^{*}17] we distinguish walking, steering, selection-based travel, and manipulation-based travel in the reviewed articles.

UrbanRama is a fully immersive urban visualization where the environment is wrapped upwards onto a cylindrical projection mixing local and global views [CMF*22]. Beyond walking and fly-based steering, also present in other immersive solutions [EPTD12, BG19], they offer a sling-shot transition after users select a building in the distance or on a map metaphor. Systems dealing with view analysis typically provide selectionbased navigation to put users into the perspective of a specific view point [DFL*15, FLD*15, OSS*17]. Pasewaldt et al. juxtapose an unwrapped multi-perspective view and a 3D perspective view of a building to support users with bi-directional selection-based navigation and view manipulation [PTD13]. Users can select a point on the 3D façade, and the view onto the 2D surface will zoom in on that region. The selection of points on the unwrapped 2D surface will trigger respective camera transitions in the perspective view. The vast majority of contributions that feature some 3D visualization of the urban environment, be it a GIS-like application or a simulation tool, make use of manipulation-based navigation where users manually change the view position and orientation. Doing so without any guidance approaches potentially results in the aforementioned context loss and high cognitive load, particularly impacting tasks that straddle multiple scales (T3 and T4).

10.5. Visual analytics systems

A fundamental aspect of 3D urban analytics is the integration of computational models with visualization and analysis processes. In their survey, Deng et al. distinguish four levels of integration: visual analytics without models (i.e., urban analyses relying on well-designed data visualizations), post-model visual analytics (models are used to discover knowledge and visualization to present it), model-integrated visual analytics (close integration between models and visualizations), and visual analytics-assisted models (visualizations used to diagnose, adjust or improve models) [DWL*23].

Visual analysis without models is common in 3D property cadastre [SKB*13, SAL*20], where tasks often require the analysis of property data without a computational model involved. The post-model visual analytics tag applies to the vast majority of domain contributions where a simulation or model is run, and then the results are analyzed (application study) or a system is produced that allows users to explore the results. Moreover, papers from the visualization community that focus on visualization and interaction contributions often fall into this category. For instance, Vis-A-Ware, where users compare building viewpoint relationships but can neither define new viewpoints nor building candidates [OSS*17], VitalVizor which allows users to explore a city with respect to urban vitality metrics while having no influence on the metrics computation [ZY18], and The Urban Toolkit [MHN*24], a grammar-based framework for the construction of urban visual analytics tools.

Considering the model-integrated visual analytics tag, Wilson et al. present a system that allows users to interact with an engine that creates block layouts and automatically computes performance indicators such as daylight or comfort [WDDH19]. Among visualization contributions, Vanegas et al. support city development with an inverse design approach to adapt city layouts to match constraints, such as lowering the average distance to the nearest park [VGDA*12]. Another design example comes from Doraiswamy et al., where users can pick and steer tower designs depending on the resulting view quality and in adherence to floor programs [DFL*15]. With Urbane [FLD*15] and Shadow Profiler [MDL*19], it is possible to interactively evaluate and compare new building developments depending on their view and shadow impact, respectively. Zhang et al. address a similar use case by letting users position, rotate, and scale a candidate building while optimal viewpoints for navigation and visibility metrics are calculated. In the domain of flood management, several works feature tight integration with a simulation model [ZZL21]. Ribicic et al. use linked scatterplots, bar charts, and a PCP to explore and evaluate barrier performance with respect to multiple simulation runs [RWF*13]. Barrier plans can be updated and instantly reevaluated. Waser et al. extend this work to scale well with a large number of flooding scenarios and respective prevention plans [WKS*14].

Works with the visual analytics-assisted models include Konev et al. [KWS*14]. Their system RunWatchers monitors, terminates, and starts new flood simulation runs from existing ones according to domain-specific rules. This leads to a more efficient traversal of the search space and improvement of overall runtimes due to the early termination of failed runs. With the same tag, Lyu et al. recently proposed a visual analytics framework for fair

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd. urban planning [LLL*23]; the framework uses a 3D map to facilitate the understanding of spatial context, including building shapes and heights, as well as to increase immersion at street level.

19 of 31

10.6. Display modalities

The vast majority of surveyed solutions run in desktop environments, with domain-specific works using a combination of desktop-based GIS tools for visualization (e.g., ArcGIS, Rhino, SAGA, Cesium ion), backed up by specific libraries (e.g., PySolar, LIS Pro 3D) for analyses and standard databases (e.g., PostGIS) for data management. Considering the visualization papers, their contributions rely on custom-built frameworks that leverage low-level libraries (e.g., OpenGL, WebGL) or game engines (e.g., Unity).

We have also found contributions using alternative display modalities. To convey 3D phenomena or urban layouts, users can be immersed in a virtual cityscape in the form of a CAVE or VR setup. Engel et al. present a 360-degree urban scene from the pedestrian level in a CAVE to support collaboration and decision-making when dealing with complex spatial information [EPTD12]. Perhac et al. present a system with VR capabilities for the visualization of social media data [PZA*17], primarily focusing on its use by urban planners. Bartosh et al. take the concept of a classical GIS system and translate it to VR [BG19]. Users can inspect layers encoding thematic information from a literal bird's eye view or inspect buildings from a pedestrian level. Building details are displayed with abstract charts in the form of a virtual fact sheet. Chen et al. do not investigate thematic data visualization but offer a deformation technique and interaction metaphors to seamlessly explore local and global urban views [CMF*22]. Chen et al. introduce an exploded view technique for immersive urban environments to deal with occlusion problems [CQW17]. Koch et al. argue that an immersive environment is suitable to illustrate complex wind flow behavior in combination with temperature and to convey the impacts of urban developments to the public [KBHS18]. In their visualization, they show 3D color-coded streamlines and a color layer on the ground encoding temperature and velocity. Zhang et al. use a VR environment to let users intuitively place and manipulate a candidate building [ZZL21]. Apart from computed performance indicators, users can also view candidates from a pedestrian level and qualitatively judge their impact on the cityscape. In the context of urban search and rescue scenarios, Bock et al. offer incident commanders an immersive view of collapsed buildings to judge the size of rescue paths [BSK*17]. We only found two systems for mobile use. XEarth translates a classical GIS application to a mobile display [LLH*15], while Chen et al. provide fire brigades en-route with in-situ information, for instance, where and how to place ladder trucks to access fire sources [CWSC14]. The exploded view technique by Chen et al. presents the only contribution using AR technology [CQW17]. The low number of immersive and in-situ contributions may indicate a research gap, however, it is likely that such systems are mostly published in more modality-specific venues (e.g., IEEE VR, ISMAR).

10.7. Evaluation methods

In our literature review, we coded the following four evaluation types: Case studies, where the authors evaluate their contri-

Miranda et al. / The State of the Art in Visual Analytics for 3D Urban Data



Figure 15: Distribution of papers according to Paper type and What dimensions, with shades denoting tag occurrence. Data creation and application study refer to domain-specific papers.

butions by presenting a use case, typically using real data without involving test subjects. Case studies are the dominant evaluation method among domain papers, especially of the domain-specific types data creation and application study. Some of theses works also introduce new methods. In this case, authors typically choose to perform a statistical evaluation in addition to the case study. Statistical evaluation describes a form of automated evaluation and validation against a baseline or ground truth without involving human test subjects, for instance in the form of validating a CFD simulation against ground truth from wind tunnel experiments [ZKL*21].

In a user study, participants are enlisted to execute a sequence of tasks in a controlled testing environment. Results can be quantitative or qualitative, such as measuring time or accuracy regarding tasks or asking their opinion about usability via Likert scale, respectively. In the surveyed papers, user studies are often used to evaluate a specific visualization technique [HWC*09, HWAT13, DZMQ16, CMF*22, BJK*22] or system [BSK*17, ZZL21]. Herbert and Chen [HC15], Seipel et al. [SAL*20], and Mota et al. [MFS*23] are examples of studies that evaluate aspects of 3D urban visualization (visualization properties, rendering attributes, and embedded plots, respectively).

Finally, expert interviews are evaluations where domain experts (and users of the software) give feedback usually in the form of structured interviews. Compared to user studies, participants are fewer and the feedback is rather interpreted than quantified. Combining a case study with expert interviews and/or a user study is the dominant evaluation pattern for design study and system papers from visualization venues. System contributions presented in domain-specific venues rarely use expert feedback with two exceptions [ABLD*22,EPTD12]. Li et al. [LLH*15], and Lv et al. [LLZ*16] conduct small-scale user studies to evaluate interactions in their GIS applications.

11. Discussion, research directions & opportunities

Despite all the surveyed contributions, 3D urban visual analytics is still incipient, especially compared to its 2D counterpart. In fact, the design of systems for 3D urban data analytics presents a large number of research opportunities in different areas, which we discuss next. In this section, we discuss a series of takeaways from our survey, focusing on visualization researchers. To help this discussion, Figure 15 shows the distribution of papers according to paper type.

11.1. Visual metaphors for thematic dimensions

While the acquisition technologies and domain analyses are becoming more sophisticated at a fast pace, the visual metaphors used to support visual analysis in 3D environments have not progressed accordingly. The majority of works rely on superimposition and linked view approaches to visualize 3D urban data. From a technological point of view, these are the most straightforward ways to integrate physical and thematic data. However, embedded views have been shown to improve user performance in certain analytical tasks when considering univariate data [MFS*23].

Common analytical scenarios. Typical applications in urban planning involve not only studying the current state of a city but also analyzing possible scenarios for development (scenario planning) that help business decisions and policy making. In such situations, thematic dimensions are often used to visually represent the criteria used to assess the quality of the physical elements (present or future) in the city. Designing visual metaphors that enable the exploration of these thematic dimensions in their spatial context is not a trivial task. For example, given the complexity of cities, this can result in complex visualization scenarios such as multivariate [DFL*15] or temporal analysis [MFS*23], these were common in a large number of surveyed works. In addition, these thematic dimensions have to be explored in multiple resolutions. For example, domain experts who analyze the quality of views along the surface of a building may be interested in analyzing this quality over a high-resolution grid or by semantic elements like floors, façades, or even for the entire building. In such cases, visual representations that depict the distribution of aggregated values are desired [EF10, SVS*22] but have not yet been applied in the 3D urban context. In summary, there is a pressing need for studies that design (possibly extending visual metaphors used in 2D) and evaluate visual metaphors for multi-variate and temporal data in 3D urban environments. This will enable a better support for recurring analytical scenarios in important application domains.

Uncertainty. The thematic dimensions used in the analysis of scenario planning are commonly the result of geometric computations or statistical modeling and, therefore, are naturally subjected to inaccuracies and uncertainty. While being of great importance for spatial analysis in general [SJ19] and urban analysis in specific (T6), almost no attention has been given to the design of new metaphors for the visualization of uncertainty for 3D urban data. Therefore, there is a need to investigate new designs to visualize uncertainty within an urban environment, as well as a more cohesive integration between urban visualization and works tackling uncertainty in information visualization [KDJ*21]. Another important direction is the design of visualizations that allow for the analysis of the interplay between model uncertainty and spatial context, particularly for building energy modeling, in which neighboring buildings play an important role in modeling accuracy and results [WYX*22]. There is also a need for work that provides a more grounded understanding of how different urban stakeholders (not only researchers but also policymakers) consider data uncertainties within the urban environment, and, if they do

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd. not consider it, understand their reasons [Hul20]. While previous work has explored how multiple data-heavy domains deal with the different aspects of uncertainty [BPHE17], no work has characterized how urban visualizations workflows are influenced by uncertainty and, especially, their propagation and impact on decisionmaking and policy. This becomes even more pressing given the implications of the data on public policies and community participation [LTM19, DWY19, HCL*20, BSMPG*22, WFC*22].

11.2. Data handling

Data formats. In our survey, we have found that there is a great diversity of formats and standards for 3D city geometry and thematic data. While some local governments make buildings' geometry available (either as meshes or XML-based formats such as CityGML) [LSB23], studies often rely on crowdsourced data such as OpenStreetMap for a description of the built environment [BCL23]. Such data requires parsers to handle missing data and complex geometries, and translating the data from or to CityGML might lead to inconsistencies. This multitude of formats and standards leads to ad-hoc data management solutions and hinders the investigation of new visual queries that integrate semantic (i.e., structural) elements of the built environment with thematic data for exploratory analysis.

Data management for interactive visualization. One important aspect of interactive data analytics is the fact that the latency experience from the interaction to the final visualization is important for the user experience and overall data exploration effectiveness [LH14]. For this reason, a large number of works have devoted attention to designing data handling strategies to support interactive exploration of large datasets [DTZM*18, BS21]. In particular, a large body of this work was designed to support the exploration of spatiotemporal data, where the spatial component is 2D. These strategies support operations such as filtering and hierarchical aggregation of thematic data for large datasets. However, these solutions have not been extended to 3D spatial data. For this reason, current 3D urban analytics systems use ad hoc data handling solutions, which often limit their data exploration capabilities and/or their performance. With the increase in data availability, there is a strong need for solutions that could support interactive explorations of 3D urban data. In particular, such a solution has to be expressive to allow for queries that align with the semantic elements in a city: buildings, façades, floors, windows, streets, neighborhoods, etc. Also, it has to be efficient to support the interactive navigation, filtering, and change in the aggregation of the data. Furthermore, this solution needs to allow for the generation and indexing of varied thematic dimensions, which is of great importance for scenario planning [FLD*15, DFL*15].

11.3. Navigation and guided explorations

Navigation. As the scale of data grows (larger spatial coverage and/or higher resolution), there is a need for innovative mechanisms to guide data exploration and visualization, with a particular emphasis on accommodating the complexities of 3D interactions. This is especially true when considering more complex scenarios such as temporal or multi-variate thematic data and highly occluded dense

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd. environments (e.g., large urban centers). In these cases, interesting urban patterns can happen in multiple (spatial/temporal) scales, and the process of searching through it by basic pan, zoom, and filtering can be ineffective. For navigation, most of the surveyed papers rely on standard 3D manipulations, with no domain-specific paper making use of walking, selection, or steering. We have also not found any sunlight access paper using walking as navigation, despite shadow playing an important role for the comfort of pedestrians [MSHC16]. Moreover, surveyed works use standard techniques in which the users have direct control over the navigation process. A promising avenue for future research involves the exploration of sketch-based navigation coupled with analytics within an urban environment. This could involve, for example, investigating of the applicability of Hagedorn and Döllner's sketch-based navigation [HD08] or Danyluk et al.'s recent proposal [DJW19] for the analysis of data within 3D urban environments.

Guided explorations and occlusion handling. Methods that help users automatically or semi-automatically select points of view or data slices to inspect, can be effective in helping users to find interesting patterns. In the case of 2D maps, there have been several techniques that help users to target specific portions of the thematic data, i.e., help to locate interesting points in space and time for further inspection. For example, the techniques proposed by Doraiswamy et al. [DFD*14], Valdivia et al. [VDP*15], and Liu et al. [LXR19] perform spatiotemporal pattern mining based on computational topology, graph signal processing, and tensor decomposition, respectively. In the context of scientific visualization, multiple works have proposed approaches to facilitate the exploration of volumetric data [BAAK*13, GWP*16, PSY*20]. Ortner et al. [OSP*16] propose a methodological framework on the integration of 3D spatial and non-spatial visualizations. However, they do not offer concrete implementations tailored to 3D urban geometry nor the integration of temporal data. Such techniques have not been applied in the context of 3D urban visualization systems. One important aspect to notice is that their computational costs and their integration with the other elements in the visual interface of a system present significant challenges.

In the case of occlusion handling, bird's view is still the most popular approach in the surveyed works across all use cases, which poses a problem given the impact that occlusion has in the different tasks highlighted in the survey. None of the domain papers make use of more complex occlusion handling techniques, such as distortion, multi-view, or steering. Automated techniques exist to solve the related problems of disocclusion [DZMQ16], viewpoint management, and route design [NPPB19]. However, these techniques try to select the best viewpoints of a target object once this object is identified. For these reasons, these have often been used to only target physical elements of the urban environment, such as landmarks and street paths. We envision that these could be coupled with techniques for pattern mining, such as the ones cited above, to provide thematic data-based occlusion handling and navigation. This will help users reduce the burden of excessive interaction to focus on spatial patterns while preserving the perception of the spatial context and facilitating tasks that straddle multiple scopes (T3 and T4).

11.4. Empirical validation of visual designs

While many 3D urban visual analytics systems have been proposed, few empirical studies were done in these scenarios to evaluate the visualizations, interactions and the overall human experience and exposure within urban spaces (see Section 10.7). This oversight suggests a slight potential disconnect with the latest human-centric developments in digital twins and geospatial science [RS19, LZL*23]. As discussed throughout this survey, the 3D urban environment adds a great deal of complexity to the navigation, interactions, and occlusion handling. While some evaluations were done in each of these cases, they are often explored in isolation [CMF*22]. This means that understanding how effective these techniques will be for data analytics is largely an open problem. Furthermore, as seen in Section 6, the vast majority of the works concentrate on the development of the systems and on the application domain and not on the evaluation of the visual designs used. This is especially critical when considering the inclusivity and accessibility requirements of various user groups. Creating inclusive visual analytics systems calls for recognizing and addressing the unique challenges faced by individuals with varying abilities (e.g., visual impairments) that may result in considerations of alternative modes of presentation, visualization, navigation, and interaction. As a result of this, the visualization metaphors and interactions used are, in many cases, done without clear guidance [NPPB19] or, at best, based on principles and studies done in 2D scenarios.

However, not all the observations concerning effective visualization designs done in the 2D spatial case can be carried over into the 3D environment [MFS*23]. Therefore, there exists a large number of opportunities to assess the current approaches and set the foundations that will guide the visual design of 3D urban visual analytics systems. In particular, we would like to highlight the great number of opportunities for evaluations concerning the use of visual metaphors and guided explorations described above. Furthermore, another important aspect that should be further explored is the interplay between occlusion handling techniques and data analytics. In fact, strategies like ghosting and deformation are good examples of techniques that fight occlusion by changing visual elements (e.g., shape, position, or transparency) of the physical elements of the scene. However, these visual elements are often used to depict thematic data (e.g., color, area). There is a need to understand what is the impact of the use of such techniques on the overall analytics of thematic data and in which situations each technique is most appropriate. Finally, many applications, especially the ones focusing on immersive experiences, try to achieve this via rendering realism. However, similar to the discussion above, the presence of rendering realism (shading, shadows) might interfere with the perception of thematic information. Again, studying the user's perception in such cases is an interesting direction for future research.

11.5. Open 3D urban visual analytics

As we reviewed and categorized papers, our immediate finding was that, while urban visualization in general has made significant advancements in previous years, there is still a notable gap between visualization contributions and domain requirements. The status quo is one where visualization contributions using 3D urban data are, more often than not, systems that are rarely used beyond the scope of the collaboration that motivated the project in the first place. In the absence of easy-to-use tools designed with visualization best practices in mind, domain experts resort to using an amalgamation of general tools and libraries, from computational notebooks (e.g., Jupyter Notebooks) to comprehensive GIS tools (e.g., ArcGIS) [YJB22]. While this approach might be sufficient to tackle well-defined urban problems in selected regions, the unprecedented challenges facing cities today call for tools and systems that lower the barrier to performing analyses that are reproducible, replicable, and extensible to enable transdisciplinary and convergent research [RBC*18, LAAD*21]. Moreover, GIS tools do not consider recent design and evaluation studies and do not provide ways to integrate novel techniques found in visualization studies, hence primarily relying on standard visualization encodings. If done right, visualization and visual analytics can play a key role in creating the needed collaborative environments that can be translated to different regions and use cases. To achieve this, there is a need for (1) community building activities that bring together urban experts and visualization researchers, and (2) the adoption of open-source practices by visualization researchers.

To bring together different communities, we believe that the roadmap outlined by Lan et al. [LYA*21] for astrophysics might be appropriate for the urban domain. This path involves joint workshops between visualization researchers and urban experts to be held at conferences in the respective areas, such as IEEE VIS and SimAUD. Examples that can serve as a guideline include ASSETS' UrbanAccess [FEH*22], and VIS' CityVis [GMB*21] and EnergyVis workshops. Another potential activity is the creation of visualization challenges using 3D urban data created by experts.

Regarding the adoption of open-source practices, in our survey, none of the visualization contributions were made publicly available, let alone built a community of contributors. In a recent survey by Yap et al. [YJB22], out of 70 open-source tools for urban analytics, only one came from the visualization community [MDL*17], and it did not take into account 3D urban data. Adjacent urban topics can provide inspiring examples to the visualization community on how to build and sustain open urban tools. Specifically, OSMnx is a widely used Python-based tool for street network analysis [Boe20]. The visualization community should consider that it is not enough to build tools, but that the scientific community should strive for public reusable tools and workflows. 3D urban visual analytics, much like urban analytics in general, is inherently complex, as the data is oftentimes large, interactions are difficult (especially in 3D), and studies are hard to translate to other regions. While reusability does not seem to be at the forefront of initiatives in the visualization community, well-engineered open efforts, accompanied by outreach efforts, are poised to be embraced by urban experts, given the variability of use cases, data, and tasks. Providing the necessary code base to facilitate urban projects in this domain can lead to not only important domain contributions but also to efforts that provide a more grounded understanding of how experts analyze and visualize 3D urban data.

11.6. Representations and visual modalities

When it comes to data visualization, the prevalence of 3D representations across examined use cases comes as little surprise. 3D representations provide better contextualization and spatial understanding compared to 2D representations. In particular, such representations allow users to better visualize how complex urban data and the related information interacts with the 3D landscape, especially in relation to the terrain, buildings or other features present in the environment. Only a handful of works utilize immersive display modalities, but almost all of them make use of a pedestrian view. Immersive environments provide a 3D spatial context that can closely resemble real-world interaction. This allows users to perceive and understand urban space more naturally, which can be beneficial for planning scenarios. Another path is utilizing immersive environments to visualize and analyze complex spatial phenomena, as for instance, to make sense of micro climate and wind simulations and putting thematic data into spatial context. However, such immersive 3D urban visual analytics solutions require carefully designed interaction and navigation metaphors as opposed to the pedestrian view. A promising opportunity to bring thematic data or planning situations directly into the actual urban space is by utilizing augmented reality glasses. The XR community has made essential progress in the last decade, such as providing accurate outdoor tracking or dealing with limited hardware capabilities of edge devices.

11.7. Collaborative urban analytics platforms

The process of enabling collaborative urban planning and assessment is pivotal for facilitating effective decision- and policymaking, while also encouraging active community involvement in the urban development processes. However, a limited number of existing urban analytics systems provide a platform for multiple stakeholders to collectively visualize, interact with, and contribute to proposed urban developments and policies. Even among the existing systems that do facilitate such collaboration, the emphasis is often placed on tailoring visualization and interaction features primarily for domain experts and practitioners, with less attention being given to incorporating the requirements of the broader public. However, there is a growing need to integrate these systems into a more inclusive context that caters to the diverse technological skills and information needs of diverse individuals. This will help bridge the gap between expert-driven functionalities and the broader public's participation and will lead to the democratization of urban exploration and planning.

12. Conclusions

In this survey, we have presented a detailed review of publications in the area of 3D urban data and visual analytics. To ensure a diverse corpus of papers, we have engaged in discussions with domain experts, also co-authors of the survey to select appropriate venues in their respective fields of study. Our selection process was then based on an extensive review of almost 20 venues, including visualization and cross-cutting ones. From an initial set of 669 papers, using a voting process where each paper was reviewed by two authors of the survey, we created a corpus of 175 papers covering a period of fifteen years. To highlight common practices and research opportunities, we framed the survey around three main questions: *Why* is 3D urban data being analyzed, *What* data is being analyzed, and *How* it is being analyzed and visualized. For each one of these

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd. questions, we have derived a set of tags covering use cases, analysis actions and targets, physical and thematic aspects of the data, and visualization and interaction. The most immediate observation is the growth in the number of publications on the topic; almost half of the surveyed works were published in the last four years. More than that, they cover a wide breadth of visualization contributions, including systems, techniques, design studies, and evaluations.

While previous works have proposed design spaces for immersive urban analytics [CWS*17], primarily focusing on VR, the same cannot be said about general 3D urban data visual analytics. Contributions to the topic are then mostly driven not by the need to explore a design space but by one-off collaborations between visualization researchers and domain experts. This hampers not only the evaluation of different visualization contributions but also the translation of research outcomes to actual workflows across different use cases. On top of that, there is a lack of toolkits specifically designed with 3D urban environments in mind. Such toolkits would also facilitate analyses across multiple scales. By considering urban studies, we have found that domain experts oftentimes need to use multiple computational frameworks and libraries to work with data and perform analyses at different scales, increasing data interoperability problems and hampering the analysis workflow. Moreover, while commercial frameworks, like ArcGIS, offer incredible capabilities for different use cases, they are closed systems, making it hard for visualization researchers to translate their contributions to scientific and engineering domains and workflows. New visualization contributions to the topic are then required to "reinvent the wheel" every time a new problem or use case is tackled - an incredibly wasteful endeavor given the data and engineering challenges that need to be solved to build a reliable tool. In this survey, we hope to have established, through a common categorization of papers and detailed discussion, a step towards driving research in visual analytics of 3D urban data. Specifically, we hope that the map of visualization and domain contributions can be used by researchers, domain experts, and practitioners as a guideline for new research opportunities.

Acknowledgments

We would like to thank the reviewers for their constructive comments and feedback. This study was supported by the Discovery Partners Institute, National Science Foundation (#2320261, #2330565), CNPq (316963/2021-6, 311425/2023-2), and FAPERJ (E-26/202.915/2019, E-26/211.134/2019). VRVis is funded by BMK, BMDW, Styria, SFG (Steirische Wirtschaftsförderungsgesellschaft m.b.H. SFG), and Vienna Business Agency in the scope of COMET-Competence Centers for Excellent Technologies (879730, 904918), managed by FFG (Österreichische Forschungsförderungsgesellschaft). This research is part of the project Multiscale Digital Twins for the Urban Environment: From Heartbeats to Cities, which is supported by the Singapore Ministry of Education Academic Research Fund Tier 1. Claudio Silva is partially supported by NASA, the National Science Foundation, DARPA, and a gift from Adobe. Any opinions, findings, and conclusions or recommendations expressed in this material are those of the authors and do not necessarily reflect the views of Adobe, DARPA, NSF, or NASA. Copyright for the reprinted images resides with the authors or publishers; images are reprinted with their permission.

References

- [AA19] ABBASABADI N., ASHAYERI M.: Urban energy use modeling methods and tools: A review and an outlook. *Building and Environment* 161 (2019), 106270. 8
- [AAA*17] ALBEAIK S., ALRISHED M., ALDAWOOD S., ALSUBAIEE S., ALFARIS A.: Virtual cities: 3D urban modeling from low resolution LiDAR data. In Proc. of the 2017 ACM SIGSPATIAL Int. Conf. on Advances in Geographic Information Systems (Redondo Beach, USA, 2017), SIGSPATIAL '17, pp. 1–4. 6, 8
- [AAA*19] ABBASABADI N., ASHAYERI M., AZARI R., STEPHENS B., HEIDARINEJAD M.: An integrated data-driven framework for urban energy use modeling (UEUM). *Applied energy 253* (2019), 113550. 8, 11
- [ABB*21] ARCAUTE E., BARTHELEMY M., BATTY M., CALDARELLI G., GERSHENSON C., HELBING D., MORENO Y., RAMASCO J., ROZENBLAT C., SÁNCHEZ A.: Future Cities: Why Digital Twins Need to Take Complexity Science on Board. Tech. rep., 2021. 2
- [ABLD*22] ANG Y., BERZOLLA Z., LETELLIER-DUCHESNE S., JUS-IEGA V., REINHART C.: UBEM.io: A web-based framework to rapidly generate urban building energy models for carbon reduction technology pathways. Sustainable Cities and Society 77 (2022), 103534. 8, 11, 14, 20
- [AC17] ALLEGRINI J., CARMELIET J.: Coupled CFD and building energy simulations for studying the impacts of building height topology and buoyancy on local urban microclimates. *Urban Climate 21* (2017), 278–305. 11
- [AC18] ALLEGRINI J., CARMELIET J.: Simulations of local heat islands in Zürich with coupled CFD and building energy models. Urban Climate 24 (2018), 340–359. 11, 15
- [AKH19] AVINI R., KUMAR P., HUGHES S.: Wind loading on high-rise buildings and the comfort effects on the occupants. Sustainable Cities and Society 45 (2019), 378–394. 7, 10, 14, 15, 16
- [Ath22] ATHAMENA K.: Microclimatic coupling to assess the impact of crossing urban form on outdoor thermal comfort in temperate oceanic climate. Urban Climate 42 (2022), 101093. 15
- [BAAK*13] BEYER J., AL-AWAMI A., KASTHURI N., LICHTMAN J. W., PFISTER H., HADWIGER M.: ConnectomeExplorer: Query-guided visual analysis of large volumetric neuroscience data. *IEEE Trans. on Visualization and Computer Graphics 19*, 12 (2013), 2868–2877. 21
- [Bat18] BATTY M.: Digital twins. Environment and Planning B: Urban Analytics and City Science 45, 5 (2018), 817–820. 2
- [BC07] BURIGAT S., CHITTARO L.: Navigation in 3D virtual environments: Effects of user experience and location-pointing navigation aids. *International Journal of Human-Computer Studies* 65, 11 (2007), 945– 958. 1
- [BC22] BILJECKI F., CHOW Y. S.: Global Building Morphology Indicators. Computers, Environment and Urban Systems 95 (2022), 101809. 2
- [BCC*18] BOUNCEUR A., CLAVIER L., COMBEAU P., MARC O., VAUZELLE R., MASSERANN A., SOLER J., EULER R., ALWAJEEH T., DEVENDRA V., NOREEN U., SORET E., LOUNIS M.: CupCarbon: A new platform for the design, simulation and 2D/3D visualization of radio propagation and interferences in IoT networks. In 2018 IEEE Annual Consumer Communications and Networking Conf. (Las Vegas, USA, 2018), CCNC '18, pp. 1–4. 8, 9, 12, 15
- [BCL23] BILJECKI F., CHOW Y. S., LEE K.: Quality of crowdsourced geospatial building information: A global assessment of OpenStreetMap attributes. *Building and Environment 237* (2023), 110295. 21
- [BDW*08] BUTKIEWICZ T., DOU W., WARTELL Z., RIBARSKY W., CHANG R.: Multi-focused geospatial analysis using probes. *IEEE Trans.* on Visualization and Computer Graphics 14, 6 (2008), 1165–1172. 8, 16

- [BG19] BARTOSH A., GU R.: Immersive representation of urban data. In Proc. of the 2019 Symposium on Simulation for Architecture and Urban Design (San Diego, USA, 2019), SimAUD '19, pp. 1–4. 8, 11, 18, 19
- [BH10] BRONZAFT A. L., HAGLER L.: Noise: The invisible pollutant that cannot be ignored. In *Emerging Environmental Technologies*, vol. 2. Springer Netherlands, 2010, pp. 75–96. 8
- [BI21] BILJECKI F., ITO K.: Street view imagery in urban analytics and GIS: A review. Landscape and Urban Planning 215 (2021), 104217. 10
- [BJK*22] BERAN D., JEDLIČKA K., KUMAR K., POPELKA S., STOTER J.: The third dimension in noise visualization – a design of new methods for continuous phenomenon visualization. *The Cartographic Journal* 59, 1 (2022), 1–17. 8, 15, 16, 18, 20
- [BK19] BONCZAK B., KONTOKOSTA C.: Large-scale parameterization of 3D building morphology in complex urban landscapes using aerial Li-DAR and city administrative data. *Computers, Environment and Urban Systems* 73 (2019), 126–142. 11, 16
- [BKH*23] BOORBOOR S., KIM Y., HU P., MOSES J. M., COLLE B. A., KAUFMAN A. E.: Submerse: Visualizing storm surge flooding simulations in immersive display ecologies. *IEEE Trans. on Visualization and Computer Graphics* (2023), 1–13. 7
- [BLSV16] BILJECKI F., LEDOUX H., STOTER J., VOSSELMAN G.: The variants of an LOD of a 3D building model and their influence on spatial analyses. *ISPRS Journal of Photogrammetry and Remote Sensing 116* (2016), 42–54. 6
- [BM13] BREHMER M., MUNZNER T.: A multi-level typology of abstract visualization tasks. *IEEE Trans. on Visualization and Computer Graphics 19*, 12 (2013), 2376–2385. 8
- [BMW*16] BREMER M., MAYR A., WICHMANN V., SCHMIDTNER K., RUTZINGER M.: A new multi-scale 3D-GIS-approach for the assessment and dissemination of solar income of digital city models. *Comput*ers, Environment and Urban Systems 57 (2016), 144–154. 6, 7
- [Boe20] BOEING G.: The right tools for the job: The case for spatial science tool-building. *Trans. in GIS 24*, 5 (2020), 1299–1314. 22
- [BPHE17] BOUKHELIFA N., PERRIN M.-E., HURON S., EAGAN J.: How data workers cope with uncertainty: A task characterisation study. In *Proc. of the 2017 CHI Conf. on Human Factors in Computing Systems* (Denver, Colorado, USA, 2017), CHI '17, Association for Computing Machinery, pp. 3645—3656. 21
- [BPMW18] BRASEBIN M., PERRET J., MUSTIÈRE S., WEBER C.: 3D urban data to assess local urban regulation influence. *Computers, Envi*ronment and Urban Systems 68 (2018), 37–52. 2
- [BRV*17] BERGER C., ROSENTRETER J., VOLTERSEN M., BAUM-GART C., SCHMULLIUS C., HESE S.: Spatio-temporal analysis of the relationship between 2D/3D urban site characteristics and land surface temperature. *Remote Sensing of Environment 193* (2017), 225–243. 2
- [BS21] BATTLE L., SCHEIDEGGER C.: A structured review of data management technology for interactive visualization and analysis. *IEEE Trans. on Visualization and Computer Graphics* 27, 2 (2021), 1128– 1138. 21
- [BSK*17] BOCK A., SVENSSON R., KLEINER A., LUNDBERG J., ROPINSKI T.: A visualization-based analysis system for urban search & rescue mission planning support. *Computer Graphics Forum 36*, 6 (2017), 148–159. 7, 11, 16, 17, 19, 20
- [BSMPG*22] BOTÍN-SANABRIA D. M., MIHAITA A.-S., PEIMBERT-GARCÍA R. E., RAMÍREZ-MORENO M. A., RAMÍREZ-MENDOZA R. A., LOZOYA-SANTOS J. D. J.: Digital twin technology challenges and applications: A comprehensive review. *Remote Sensing 14*, 6 (2022). 21
- [BVR07] BRONZAFT A., VAN RYZIN G.: Neighborhood noise and its consequences. Survey Research Unit, School of Public Affairs, Baruch College, New York 2 (2007). 8
- [CBK*19] CORNEL D., BUTTINGER-KREUZHUBER A., KONEV A.,

© 2024 The Authors.

HORVÁTH Z., WIMMER M., HEIDRICH R., WASER J.: Interactive visualization of flood and heavy rain simulations. *Computer Graphics Forum* 38, 3 (2019), 25–39. 6, 7, 10, 11, 12, 15

- [CD21] CLARINVAL A., DUMAS B.: Intra-city traffic data visualization: a systematic literature review. *IEEE Trans. on Intelligent Transportation Systems* 23, 7 (2021), 6298–6315. 2
- [CFL14] CHOW A., FUNG A. S., LI S.: GIS modeling of solar neighborhood potential at a fine spatiotemporal resolution. *Buildings 4*, 2 (2014), 195–206. 15
- [CG17] CHUN B., GUHATHAKURTA S.: Daytime and nightime urban heat islands statistical models for Atlanta. *Environment and Planning B: Urban Analytics and City Science* 44, 2 (2017), 308–327. 2
- [Cha22] CHARITONIDOU M.: Urban scale digital twins in data-driven society: Challenging digital universalism in urban planning decisionmaking. *International Journal of Architectural Computing 20*, 2 (2022), 238–253. 2
- [CKS*15] CORNEL D., KONEV A., SADRANSKY B., HORVÁTH Z., GROLLER E., WASER J.: Visualization of object-centered vulnerability to possible flood hazards. *Computer Graphics Forum 34*, 3 (2015), 331–340. 7, 9, 10, 14, 16, 17, 18
- [CLJ*19] CASAZZA M., LEGA M., JANNELLI E., MINUTILLO M., JAFFE D., SEVERINO V., ULGIATI S.: 3D monitoring and modelling of air quality for sustainable urban port planning: Review and perspectives. *Journal of Cleaner Production 231* (2019), 1342–1352. 2
- [CMF*22] CHEN S., MIRANDA F., FERREIRA N., LAGE M., DO-RAISWAMY H., BRENNER C., DEFANTI C., KOUTSOUBIS M., WIL-SON L., PERLIN K., SILVA C.: UrbanRama: Navigating cities in virtual reality. *IEEE Trans. on Visualization and Computer Graphics* 28, 12 (2022), 4685–4699. 1, 2, 9, 12, 17, 18, 19, 20, 22
- [CQW17] CHEN Z., QU H., WU Y.: Immersive urban analytics through exploded views. In Workshop on Immersive Analytics (2017). 1, 2, 10, 18, 19
- [CRPB14] CATITA C., REDWEIK P., PEREIRA J., BRITO M.: Extending solar potential analysis in buildings to vertical facades. *Computers & Geosciences 66* (2014), 1–12. 11, 16, 18
- [CW23] CABRAL A., WALDO J.: Designing large-scale wireless sensor networks for urban environmental sensing. In Adjunct Proc. of the 2023 ACM International Joint Conference on Pervasive and Ubiquitous Computing & the 2023 ACM International Symposium on Wearable Computing (2023), UbiComp/ISWC '23 Adjunct, Association for Computing Machinery, p. 253–257. 2
- [CWK*07] CHANG R., WESSEL G., KOSARA R., SAUDA E., RIB-ARSKY W.: Legible Cities: Focus-dependent multi-resolution visualization of urban relationships. *IEEE Trans. on Visualization and Computer Graphics 13*, 6 (2007), 1169–1175. 11, 16
- [CWS*17] CHEN Z., WANG Y., SUN T., GAO X., CHEN W., PAN Z., QU H., WU Y.: Exploring the design space of immersive urban analytics. *Visual Informatics* 1, 2 (2017), 132–142. 1, 2, 3, 15, 23
- [CWSC14] CHEN L.-C., WU C.-H., SHEN T.-S., CHOU C.-C.: The application of geometric network models and building information models in geospatial environments for fire-fighting simulations. *Computers, Environment and Urban Systems* 45 (2014), 1–12. 7, 19
- [DFD*14] DORAISWAMY H., FERREIRA N., DAMOULAS T., FREIRE J., SILVA C.: Using topological analysis to support event-guided exploration in urban data. *IEEE Trans. on Visualization and Computer Graphics 20*, 12 (2014), 2634–2643. 21
- [DFFL17] DROBEŽ P., FRAS M., FERLAN M., LISEC A.: Transition from 2D to 3D real property cadastre: The case of the Slovenian cadastre. *Computers, Environment and Urban Systems* 62 (2017), 125–135. 8
- [DFL*15] DORAISWAMY H., FERREIRA N., LAGE M., VO H., WIL-SON L., WERNER H., PARK M., SILVA C.: Topology-based catalogue exploration framework for identifying view-enhanced tower designs. ACM Trans. on Graphics 34, 6 (2015), 1–13. 6, 8, 9, 12, 14, 16, 17, 18, 19, 20, 21

© 2024 The Authors.

- [DFL*18] DORAISWAMY H., FREIRE J., LAGE M., MIRANDA F., SILVA C.: Spatio-temporal urban data analysis: A visual analytics perspective. *IEEE Computer Graphics and Applications* 38, 5 (2018), 26– 35. 3
- [DFT19] DANI V., FREITAS C., THOM L.: Ten years of visualization of business process models: A systematic literature review. *Computer Standards & Interfaces 66* (2019), 103347. 2
- [DJW19] DANYLUK K., JENNY B., WILLETT W.: Look-from camera control for 3D terrain maps. In *Proc. of the 2019 CHI Conf. on Human Factors in Computing Systems* (New York, NY, USA, 2019), CHI '19, Association for Computing Machinery, p. 1–12. 21
- [DL19] DE LUCA F.: Sun and wind: Integrated environmental performance analysis for building and pedestrian comfort. In *Proc. of the* 2019 Symposium on Simulation for Architecture and Urban Design (San Diego, USA, 2019), SimAUD '19, pp. 1–8. 6, 7, 9, 14
- [DLD*16] DELMAS A., LOBACCARO G., DONN M., MUSY M., GARDE F.: Solar potential in extreme climate conditions: comparative analysis of two district case studies in Norway and Reunion Island. In Proc. of the 2016 Symposium on Simulation for Architecture and Urban Design (London, UK, 2016), SimAUD '16. 6, 9, 10
- [DLV17] DE LUCA F., VOLL H.: Computational method for variable objectives and context aware solar envelopes generation. In *Proc. of the* 2017 Symposium on Simulation for Architecture and Urban Design (San Diego, USA, 2017), SimAUD '17, pp. 1–8. 6, 10, 12
- [DRST14] DÜBEL S., RÖHLIG M., SCHUMANN H., TRAPP M.: 2D and 3D presentation of spatial data: A systematic review. In 2014 IEEE VIS International Workshop on 3DVis (3DVis) (2014), pp. 11–18. 16
- [DTZM*18] DORAISWAMY H., TZIRITA ZACHARATOU E., MIRANDA F., LAGE M., AILAMAKI A., SILVA C. T., FREIRE J.: Interactive visual exploration of spatio-temporal urban data sets using Urbane. In *Proc. of the 2018 ACM Int. Conf. on Management of Data* (Houston, USA, 2018), SIGMOD '18, pp. 1693–1696. 21
- [DvdGP*20] DEININGER M., VON DER GRÜN M., PIEPEREIT R., SCHNEIDER S., SANTHANAVANICH T., COORS V., VOSS U.: A continuous, semi-automated workflow: From 3D city models with geometric optimization and CFD simulations to visualization of wind in an urban environment. *ISPRS International Journal of Geo-Information 9*, 11 (2020), 657. 2
- [DWL*20] DEMBSKI F., WÖSSNER U., LETZGUS M., RUDDAT M., YAMU C.: Urban digital twins for smart cities and citizens: The case study of Herrenberg, Germany. *Sustainability* 12, 6 (2020), 2307. 2
- [DWL*23] DENG Z., WENG D., LIU S., TIAN Y., XU M., WU Y.: A survey of urban visual analytics: Advances and future directions. *Computational Visual Media* 9, 1 (2023), 3–39. 3, 19
- [DWY19] DEMBSKI F., WÖSSNER U., YAMU C.: Virtual reality and space syntax: Civic engagement and decision support for smart, sustainable cities. In Proc. of the 2019 Int. Space Syntax Conf. (Beijing, China, 2019), pp. 8–13. 2, 21
- [DYW*19] DISSEGNA M., YIN T., WEI S., RICHARDS D., GRÊT-REGAMEY A.: 3-D reconstruction of an urban landscape to assess the influence of vegetation in the radiative budget. *Forests 10*, 8 (2019), 700. 2
- [DZD*10] DRATVA J., ZEMP E., DIETRICH D., BRIDEVAUX P.-O., ROCHAT T., SCHINDLER C., GERBASE M.: Impact of road traffic noise annoyance on health-related quality of life: results from a populationbased study. *Quality of Life Research 19*, 1 (2010), 37–46. 8
- [DZMQ16] DENG H., ZHANG L., MAO X., QU H.: Interactive urban context-aware visualization via multiple disocclusion operators. *IEEE Trans. on Visualization and Computer Graphics* 22, 7 (2016), 1862– 1874. 7, 18, 20, 21
- [EF10] ELMQVIST N., FEKETE J.-D.: Hierarchical aggregation for information visualization: Overview, techniques, and design guidelines. *IEEE Trans. on Visualization and Computer Graphics* 16, 3 (2010), 439– 454. 20

- [EGAB17] ELSHAER A., GAIROLA A., ADAMEK K., BITSUAMLAK G.: Variations in wind load on tall buildings due to urban development. *Sustainable Cities and Society* 34 (2017), 264–277. 7, 10, 14, 15, 16, 17
- [Els21] ELSAYED D. S. I.: The microclimatic impacts of urban spaces on the behaviour of pandemics between propagation and containment: Case study historic Cairo. Urban Climate 36 (2021), 100773. 2, 7
- [EMM17] ESTOQUE R. C., MURAYAMA Y., MYINT S. W.: Effects of landscape composition and pattern on land surface temperature: An urban heat island study in the megacities of Southeast Asia. *Science of The Total Environment 577* (2017), 349–359. 2
- [EPTD12] ENGEL J., PASEWALDT S., TRAPP M., DÖLLNER J.: An immersive visualization system for virtual 3D city models. In *Proc. of the 2012 IEEE Int. Conf. on Geoinformatics* (Hong Kong, China, 2012), pp. 1–7. 6, 18, 19, 20
- [ET08] ELMQVIST N., TSIGAS P.: A taxonomy of 3D occlusion management for visualization. *IEEE Trans. on Visualization and Computer Graphics 14*, 5 (2008), 1095–1109. 8, 18
- [ETT07] ELMQVIST N., TUDOREANU M. E., TSIGAS P.: Tour generation for exploration of 3D virtual environments. In *Proc. of the 2007* ACM Symposium on Virtual Reality Software and Technology (Newport Beach, California, 2007), VRST '07, Association for Computing Machinery, pp. 207—210. 1
- [ETVF20] EREMCHENKO E., TIKUNOV V., VYLEGZHANIN S., FETI-SHEV A.: Infectious dynamics in urban 3D-environment: Challenges and possibilities. In Proc. of the 2020 CEUR Workshop (2020). 2
- [FEH*22] FROEHLICH J. E., EISENBERG Y., HOSSEINI M., MIRANDA F., ET AL.: The future of urban accessibility for people with disabilities: Data collection, analytics, policy, and tools. In Proc. of the 24th Int. ACM SIGACCESS Conf. on Computers and Accessibility (Athens, Greece, 2022), ASSETS '22, Association for Computing Machinery. 22
- [FH09] FUCHS R., HAUSER H.: Visualization of multi-variate scientific data. Computer Graphics Forum 28, 6 (2009), 1670–1690. 15
- [FLD*15] FERREIRA N., LAGE M., DORAISWAMY H., VO H., WIL-SON L., WERNER H., PARK M., SILVA C.: Urbane: A 3D framework to support data driven decision making in urban development. In *Proc.* of the 2015 IEEE Conf. on Visual Analytics Science and Technology (Chicago, USA, 2015), VAST '15, pp. 97–104. 8, 9, 11, 12, 14, 15, 16, 17, 18, 19, 21
- [FPP*21] FLORIO P., PERONATO G., PERERA A., DI BLASI A., POON K., KÄMPF J. H.: Designing and assessing solar energy neighborhoods from visual impact. *Sustainable Cities and Society* 71 (2021), 102959.
- [FQY*22] FENG Z., QU H., YANG S., DING Y., SONG J.: A survey of visual analytics in urban area. *Expert Systems* 39, 9 (2022), e13065. 3
- [FWC22] FIJAŁKOWSKA A., WAKSMUNDZKA K., CHMIEL J.: Assessment of the effectiveness of photovoltaic panels at public transport stops: 3D spatial analysis as a tool to strengthen decision making. *Energies 15*, 3 (2022). 6, 8, 15, 16
- [GBC20] GAUTIER J., BREDIF M., CHRISTOPHE S.: Co-visualization of air temperature and urban data for visual exploration. In *Proc. of the* 2020 IEEE Visualization Conf. (Salt Lake City, USA, 2020), VIS '20, pp. 71–75. 7, 15, 18
- [GBI14] GROS A., BOZONNET E., INARD C.: Cool materials impact at district scale—coupling building energy and microclimate models. Sustainable Cities and Society 13 (2014), 254–266. 7, 8, 16
- [GCA06] GUITE H., CLARK C., ACKRILL G.: The impact of the physical and urban environment on mental well-being. *Public Health 120*, 12 (2006), 1117–1126. 8
- [GCB20] GAUTIER J., CHRISTOPHE S., BRÉDIF M.: Visualizing 3D climate data in urban 3D models. *International Archives of the Photogrammetry, Remote Sensing & Spatial Information Sciences XLIII-B4-2020* (2020), 781–789. 2
- [GL19] GALLAGHER J., LAGO C.: How parked cars affect pollutant dispersion at street level in an urban street canyon? A CFD modelling exercise assessing geometrical detailing and pollutant decay rates. *Science* of The Total Environment 651 (2019), 2410–2418. 7

- [GLU09] GÁL T., LINDBERG F., UNGER J.: Computing continuous sky view factors using 3D urban raster and vector databases: comparison and application to urban climate. *Theoretical and Applied Climatology* 95, 1-2 (2009), 111–123. 2
- [GMB*21] GOODWIN S., MEIER S., BARTRAM L., GODWIN A., NAGEL T., DÖRK M.: Unravelling the human perspective and considerations for urban data visualization. In *Proc. of the 2021 IEEE 14th Pacific Visualization Symposium* (Tianjin, China, 2021), PacificVis '21, pp. 126–130. 22
- [GWP*16] GU Y., WANG C., PETERKA T., JACOB R., KIM S. H.: Mining graphs for understanding time-varying volumetric data. *IEEE Trans.* on Visualization and Computer Graphics 22, 1 (2016), 965–974. 21
- [HC15] HERBERT G., CHEN X.: A comparison of usefulness of 2D and 3D representations of urban planning. *Cartography and Geographic Information Science* 42, 1 (2015), 22–32. 7, 20
- [HCL*20] HONG T., CHEN Y., LUO X., LUO N., LEE S. H.: Ten questions on urban building energy modeling. *Building and Environment 168* (2020), 106508. 21
- [HD08] HAGEDORN B., DÖLLNER J.: Sketch-based navigation in 3D virtual environments. In *Smart Graphics* (Berlin, Heidelberg, 2008), Butz A., Fisher B., Krüger A., Olivier P., Christie M., (Eds.), Springer Berlin Heidelberg, pp. 239–246. 21
- [HDVT*08] HARALABIDIS A., DIMAKOPOULOU K., VIGNA-TAGLIANTI F., GIAMPAOLO M., ET AL.: Acute effects of night-time noise exposure on blood pressure in populations living near airports. *European Heart Journal 29*, 5 (2008), 658–664. 8
- [HHG23] HAERI T., HASSAN N., GHAFFARIANHOSEINI A.: Evaluation of microclimate mitigation strategies in a heterogenous street canyon in Kuala Lumpur from outdoor thermal comfort perspective using Envimet. Urban Climate 52 (2023), 101719. 11, 18
- [HM17] HANKEY S., MARSHALL J. D.: Urban form, air pollution, and health. Current environmental health reports 4 (2017), 491–503. 2
- [HP20] HADAVI M., PASDARSHAHRI H.: Quantifying impacts of wind speed and urban neighborhood layout on the infiltration rate of residential buildings. *Sustainable Cities and Society 53* (2020), 101887. 7, 12, 15
- [HP21] HADAVI M., PASDARSHAHRI H.: Impacts of urban buildings on microclimate and cooling systems efficiency: Coupled CFD and BES simulations. *Sustainable Cities and Society* 67 (2021), 102740. 7, 8, 14, 15, 16, 17
- [HSN14] HAMMER M., SWINBURN T., NEITZEL R.: Environmental noise pollution in the united states: Developing an effective public health response. *Environmental Health Perspectives 122*, 2 (2014), 115–119. 8
- [Hul20] HULLMAN J.: Why authors don't visualize uncertainty. *IEEE Trans. on Visualization and Computer Graphics* 26, 1 (2020), 130–139. 21
- [HW08] HAKLAY M., WEBER P.: OpenStreetMap: User-generated street maps. *IEEE Pervasive Computing* 7, 4 (2008), 12–18. 2, 10
- [HW19a] HU L., WENDEL J.: Analysis of urban surface morphologic effects on diurnal thermal directional anisotropy. *ISPRS Journal of Pho*togrammetry and Remote Sensing 148 (2019), 1–12. 7, 10, 16
- [HW19b] HUANG X., WANG Y.: Investigating the effects of 3D urban morphology on the surface urban heat island effect in urban functional zones by using high-resolution remote sensing data: A case study of Wuhan, Central China. *ISPRS Journal of Photogrammetry and Remote Sensing 152* (2019), 119–131. 2
- [HWAT13] HIRONO D., WU H.-Y., ARIKAWA M., TAKAHASHI S.: Constrained optimization for disoccluding geographic landmarks in 3D urban maps. In *Proc. of the 2013 IEEE Pacific Visualization Symposium* (Sydney, Australia, 2013), PacificVis '13, pp. 17–24. 18, 20
- [HWC*09] HUAMIN Q., WANG H., CUI W., WU Y., CHAN M.-Y.: Focus+context route zooming and information overlay in 3D urban environments. *IEEE Trans. on Visualization and Computer Graphics* 15, 6 (2009), 1547–1554. 18, 20

© 2024 The Authors.

4678659, 2024, 3, Downloaded from https://onlinelibrary.wiley.com/doi/10.1111/cgf.15112, Wiley Online Library on [15/06/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

- [HYF*23] HUANG C., YAO J., FU B., CALAUTIT J. K., ZHAO C., HUANG J., BAN Q.: Sensitivity analysis of WRF-CFD-based downscaling methods for evaluation of urban pedestrian-level wind. Urban Climate 49 (2023), 101569. 7, 9, 11, 14
- [HYK*21] HU Q., YANG B., KHALID S., XIAO W., TRIGONI N., MARKHAM A.: Towards semantic segmentation of urban-scale 3d point clouds: A dataset, benchmarks and challenges. In *Proc. of the IEEE/CVF Conf. on Computer Vision and Pattern Recognition* (2021). 10
- [HYK*22] HU Q., YANG B., KHALID S., XIAO W., TRIGONI N., MARKHAM A.: SensatUrban: Learning semantics from urban-scale photogrammetric point clouds. *International Journal of Computer Vision* 130, 2 (2022), 316–343. 10
- [HZL*23] HE H., ZHU Y., LIU L., DU J., LIU L., LIU J.: Effects of roadside trees three-dimensional morphology characteristics on trafficrelated PM2.5 distribution in hot-humid urban blocks. *Urban Climate* 49 (2023), 101448. 7
- [JK21] JANG G., KIM S.: Are decline-oriented strategies thermally sustainable in shrinking cities? Urban Climate 39 (2021), 100924. 14
- [JMB*22] JIANG F., MA L., BROYD T., CHEN W., LUO H.: Digital twin enabled sustainable urban road planning. *Sustainable Cities and Society* 78 (2022), 103645. 2
- [JN20] JAVANROODI K., NIK V. M.: Interactions between extreme climate and urban morphology: Investigating the evolution of extreme wind speeds from mesoscale to microscale. *Urban Climate 31* (2020), 100544. 14, 18
- [JPS*20] JOHARI F., PERONATO G., SADEGHIAN P., ZHAO X., WIDÉN J.: Urban building energy modeling: State of the art and future prospects. *Renewable and Sustainable Energy Reviews 128* (2020), 109902.
- [JRK14] JAZAYERI I., RAJABIFARD A., KALANTARI M.: A geometric and semantic evaluation of 3D data sourcing methods for land and property information. *Land Use Policy 36* (2014), 219–230. 8
- [JVHKL01] JOHNSTON K., VER HOEF J. M., KRIVORUCHKO K., LU-CAS N.: Using ArcGIS geostatistical analyst, vol. 380. Esri Redlands, 2001. 17
- [JY20] JOHANSSON E., YAHIA M. W.: Wind comfort and solar access in a coastal development in malmö, Sweden. Urban Climate 33 (2020), 100645. 7
- [KBHS18] KOCH F., BILKE L., HELBIG C., SCHLINK U.: Compact or cool? The impact of brownfield redevelopment on inner-city micro climate. *Sustainable Cities and Society* 38 (2018), 31–41. 6, 7, 14, 15, 19
- [KCG22] KABOŠOVÁ L., CHRONIS A., GALANOS T.: Fast wind prediction incorporated in urban city planning. *International Journal of Architectural Computing 20*, 3 (2022), 511–527. 11
- [KD17] KOZIATEK O., DRAGIĆEVIĆ S.: iCity 3D: A geosimulation method and tool for three-dimensional modeling of vertical urban development. *Landscape and Urban Planning 167* (2017), 356–367. 8, 11
- [KD19] KOZIATEK O., DRAGIĆEVIĆ S.: A local and regional spatial index for measuring three-dimensional urban compactness growth. *Envi*ronment and Planning B: Urban Analytics and City Science 46, 1 (2019), 143–164. 8
- [KD21] KOLBE T., DONAUBAUER A.: Semantic 3D city modeling and BIM. In Urban Informatics. Springer Singapore, 2021, pp. 609–636. 2
- [KDJ*21] KAMAL A., DHAKAL P., JAVAID A. Y., DEVABHAKTUNI V. K., KAUR D., ZAIENTZ J., MARINIER R.: Recent advances and challenges in uncertainty visualization: a survey. *Journal of Visualization* 24, 5 (2021), 861–890. 20
- [kep] kepler.gl. https://kepler.gl. Accessed: May 25, 2023. 17
- [KFS*22] KRAUS M., FUCHS J., SOMMER B., KLEIN K., ENGELKE U., KEIM D., SCHREIBER F.: Immersive analytics with abstract 3D visualizations: A survey. *Computer Graphics Forum* 41, 1 (2022), 201– 229. 3

© 2024 The Authors.

- [KK19] KRIETEMEYER B., KONTAR R.: A method for integrating an UBEM with GIS for spatiotemporal visualization and analysis. In Proc. of the 2019 Symposium on Simulation for Architecture and Urban Design (San Diego, USA, 2019), SimAUD '19. 8, 16, 17
- [KKK21] KARIMIMOSHAVER M., KHALVANDI R., KHALVANDI M.: The effect of urban morphology on heat accumulation in urban street canyons and mitigation approach. *Sustainable Cities and Society 73* (2021), 103127. 7, 11, 12, 15
- [KK018] KAYNAK S., KAYNAK B., ÖZMEN A.: A software tool development study for solar energy potential analysis. *Energy and Buildings* 162 (2018), 134–143. 6, 12
- [KM17] KAYA H. S., MUTLU H.: Modelling 3D spatial enclosure of urban open spaces. *Journal of Urban Design* 22, 1 (2017), 96–115. 8
- [KPMB21] KADAVERUGU R., PUROHIT V., MATLI C., BINIWALE R.: Improving accuracy in simulation of urban wind flows by dynamic downscaling WRF with OpenFOAM. Urban Climate 38 (2021), 100912. 7, 11, 17, 18
- [KWS*14] KONEV A., WASER J., SADRANSKY B., CORNEL D., PERDIGAO R., HORVATH Z., GROLLER M. E.: Run watchers: Automatic simulation-based decision support in flood management. *IEEE Trans. on Visualization and Computer Graphics 20*, 12 (2014), 1873– 1882. 7, 9, 19
- [LAAD*21] LOBO J., ALBERTI M., ALLEN-DUMAS M., BETTEN-COURT L. M., BEUKES A., BOJÓRQUEZ TAPIA L. A., CHEN W.-Q., DODGE A., NEAL Z., PERREIRA A., ET AL.: A convergence research perspective on graduate education for sustainable urban systems science. *npj Urban Sustainability 1*, 39 (2021). 22
- [LCL22] LV Z., CHEN D., LV H.: Smart city construction and management by digital twins and BIM big data in COVID-19 scenario. ACM Trans. on Multimedia Computing, Communications, and Applications 18, 2s (2022), 1–21. 2
- [LD10] LORENZ H., DÖLLNER J.: 3D feature surface properties and their application in geovisualization. *Computers, Environment and Ur*ban Systems 34, 6 (2010), 476–483. 7, 11, 15
- [LGL114] LIANG J., GONG J., LI W., IBRAHIM A.: A visualizationoriented 3D method for efficient computation of urban solar radiation based on 3D–2D surface mapping. *International Journal of Geographical Information Science* 28, 4 (2014), 780–798. 6
- [LH14] LIU Z., HEER J.: The effects of interactive latency on exploratory visual analysis. *IEEE Trans. on Visualization and Computer Graphics* 20, 12 (2014), 2122–2131. 21
- [Lin05] LINDBERG F.: Towards the use of local governmental 3-D data within urban climatology studies. *Mapping and Image Science* 2 (2005), 32–37. 2
- [LJKM*17] LAVIOLA JR J., KRUIJFF E., MCMAHAN R. P., BOWMAN D., POUPYREV I. P.: 3D user interfaces: theory and practice. Pearson, 2017. 18
- [LJR*14] LEE E. Y., JERRETT M., ROSS Z., COOGAN P. F., SETO E. Y.: Assessment of traffic-related noise in three cities in the united states. *Environmental Research 132* (2014), 182–189. 14, 16
- [LJSB23] LEI B., JANSSEN P., STOTER J., BILJECKI F.: Challenges of urban digital twins: A systematic review and a Delphi expert survey. *Automation in Construction 147* (2023), 104716. 2
- [LLC*20] LI Q., LIU Y. J., CHEN L., YANG X. C., PENG Y., YUAN X. R., WIJERATHNE M. L. L.: SEEVis: A smart emergency evacuation plan visualization system with data-driven shot designs. *Computer Graphics Forum 39*, 3 (2020), 523–535. 18
- [LLH*15] LI X., LV Z., HU J., ZHANG B., SHI L., FENG S.: XEarth: A 3D GIS platform for managing massive city information. In Proc. of the 2015 IEEE Int. Conf. on Computational Intelligence and Virtual Environments for Measurement Systems and Applications (Shenzhen, China, 2015), CIVEMSA '15, pp. 1–6. 19, 20

- [LLL*23] LYU Y., LU H., LEE M. K., SCHMITT G., LIM B. Y.: IF-City: Intelligible fair city planning to measure, explain and mitigate inequality. IEEE Trans. on Visualization and Computer Graphics (2023), 1-18. 19
- [LLT*19] LI L., LEI Y., TANG L., YAN F., LUO F., ZHU H.: A 3D spatial data model of the solar rights associated with individual residential properties. Computers, Environment and Urban Systems 74 (2019), 88-99. 7
- [LLZ*16] LV Z., LI X., ZHANG B., WANG W., ZHU Y., HU J., FENG S.: Managing big city information based on WebVRGIS. IEEE Access 4 (2016), 407–415. 8, 18, 20
- [LMBZ18] LLAGUNO-MUNITXA M., BOU-ZEID E.: Shaping buildings to promote street ventilation: A large-eddy simulation study. Urban Climate 26 (2018), 76-94. 11
- [LMS*23] LI Z., MING T., SHI T., ZHANG H., WEN C.-Y., WU Y., WANG C., YIN K., DE RICHTER R., LI W.: CFD modeling of traffic tidal flow: Assessment of pollutant dispersion. Urban Climate 47 (2023), 101380. 7, 11
- [LS13] LEIDI M., SCHLÜTER A.: Exploring urban space: Volumetric site analysis for conceptual design in the urban context. International Journal of Architectural Computing 11, 2 (2013), 157–182. 8
- [LSAG22] LEVINE D., SUSSMAN S., AHARON-GUTMAN M.: Spatialtemporal patterns of self-organization: A dynamic 4D model for redeveloping the post-zoning city. Environment and Planning B: Urban Analytics and City Science 49, 3 (2022), 1005-1023. 8
- [LSB23] LEI B., STOUFFS R., BILJECKI F.: Assessing and benchmarking 3D city models. International Journal of Geographical Information Science 37 (2023), 788-809. 21
- [LTM19] L. W., T. N., M. S. J.: Developing a city-level digital twin -propositions and a case study. In Int. Conf. on Smart Infrastructure and Construction 2019 (ICSIC) (2019), pp. 187-194. 2, 21
- [LX21] LI H., XIE H.: Noise exposure of the residential areas close to urban expressways in a high-rise mountainous city. Environment and Planning B: Urban Analytics and City Science 48, 6 (2021), 1414–1429. 16
- [LXR19] LIU D., XU P., REN L.: TPFlow: Progressive partition and multidimensional pattern extraction for large-scale spatio-temporal data analysis. IEEE Trans. on Visualization and Computer Graphics 25, 1 (2019), 1–11. 21
- [LXWY22] LI M., XUE F., WU Y., YEH A.: A room with a view: Automatic assessment of window views for high-rise high-density areas using city information models and deep transfer learning. Landscape and Urban Planning 226 (2022), 104505. 8, 9, 10, 11, 12, 14, 15, 16
- [LYA*21] LAN F., YOUNG M., ANDERSON L., YNNERMAN A., BOCK A., BORKIN M. A., FORBES A. G., KOLLMEIER J. A., WANG B.: Visualization in astrophysics: Developing new methods, discovering our universe, and educating the earth. Computer Graphics Forum 40, 3 (2021), 635-663. 22
- [LZL*23] LIU P., ZHAO T., LUO J., LEI B., FREI M., MILLER C., BIL-JECKI F.: Towards human-centric digital twins: Leveraging computer vision and graph models to predict outdoor comfort. Sustainable Cities and Society 93 (2023), 104480. 22
- [MDL*17] MIRANDA F., DORAISWAMY H., LAGE M., ZHAO K., GONÇALVES B., WILSON L., HSIEH M., SILVA C. T.: Urban Pulse: Capturing the rhythm of cities. IEEE Trans. on Visualization and Computer Graphics 23, 1 (2017), 791-800. 22
- [MDL*19] MIRANDA F., DORAISWAMY H., LAGE M., WILSON L., HSIEH M., SILVA C.: Shadow Accrual Maps: Efficient accumulation of city-scale shadows over time. IEEE Trans. on Visualization and Computer Graphics 25, 3 (2019), 1559-1574. 6, 9, 10, 11, 12, 14, 15, 16, 17, 19
- [MFS*23] MOTA R., FERREIRA N., SILVA J., HORGA M., LAGE M., CEFERINO L., ALIM U., SHARLIN E., MIRANDA F.: A comparison of spatiotemporal visualizations for 3D urban analytics. IEEE Trans. on

Visualization and Computer Graphics 29, 1 (2023), 1277-1287. 6, 7, 10, 12, 16, 17, 20, 22

- [MG20] MEGAHED N. A., GHONEIM E. M.: Antivirus-built environment: Lessons learned from Covid-19 pandemic. Sustainable Cities and Society 61 (2020), 102350. 2
- [MHN*24] MOREIRA G., HOSSEINI M., NIPU M. N. A., LAGE M., FERREIRA N., MIRANDA F.: The Urban Toolkit: A grammar-based framework for urban visual analytics. IEEE Trans. on Visualization and Computer Graphics (2024). 19
- [MKG21] MORRISON W., KOTTHAUS S., GRIMMOND S.: Urban surface temperature observations from ground-based thermography: intraand inter-facet variability. Urban Climate 35 (2021), 100748. 11
- [MKK*18] MIAO Y., KOENIG R., KNECHT K., KONIEVA K., BUS P., CHANG M.-C.: Computational urban design prototyping: Interactive planning synthesis methods-a case study in cape town. International Journal of Architectural Computing 16, 3 (2018), 212–226. 8
- [MnPBM*17] MUÑOZ PANDIELLA I., BOSCH C., MÉRILLOU N., PUEYO X., MÉRILLOU S.: Real-time solar exposure simulation in complex cities: Real-time solar exposure simulation in complex cities. Computer Graphics Forum 36, 8 (2017), 554-566. 6, 12, 15
- [MSD20] MOKHTAR S., SOJKA A., DAVILA C.: Conditional generative adversarial networks for pedestrian wind flow approximation. In Proc. of the 2020 Symposium on Simulation for Architecture and Urban Design (San Diego, USA, 2020), SimAUD '20, pp. 1-8. 11
- [MSHC16] MIDDEL A., SELOVER N., HAGEN B., CHHETRI N.: Impact of shade on outdoor thermal comfort-a seasonal field study in Tempe, Arizona. International Journal of Biometeorology 60, 12 (2016), 1849-1861, 21
- [MTML18] MOGHADAM S., TONIOLO J., MUTANI G., LOMBARDI P.: A GIS-statistical approach for assessing built environment energy use at urban scale. Sustainable Cities and Society 37 (2018), 70-84. 8, 9, 10, 16
- [Mun08] MUNZNER T.: Process and pitfalls in writing information visualization research papers. In Information Visualization: Human-Centered Issues and Perspectives. Springer Berlin Heidelberg, 2008, pp. 134-153. 5
- [Mun15] MUNZNER T.: Visualization analysis and design. A.K. Peters visualization series. CRC Press, 2015. 2, 8, 10
- [NBW03] NOTH M., BORNING A., WADDELL P.: An extensible, modular architecture for simulating urban development, transportation, and environmental impacts. Computers, Environment and Urban Systems 27, 2 (2003), 181–203. 17
- [NPPB19] NEUVILLE R., POULIOT J., POUX F., BILLEN R.: 3D viewpoint management and navigation in urban planning: Application to the exploratory phase. Remote Sensing 11, 3 (2019), 236. 21, 22
- [NSN18] NAZARIAN N., SIN T., NORFORD L.: Numerical modeling of outdoor thermal comfort in 3D. Urban Climate 26 (2018), 212-230. 11
- [NZCE17] NOUVEL R., ZIRAK M., COORS V., EICKER U.: The influence of data quality on urban heating demand modeling using 3D city models. Computers, Environment and Urban Systems 64 (2017), 68-80. 8, 10, 15
- [OS18] OURIA M., SEVINC H.: Evaluation of the potential of solar energy utilization in Famagusta, Cyprus. Sustainable Cities and Society 37 (2018), 189–202. 6, 8, 11, 16
- [OSP*16] ORTNER T., SORGER J., PIRINGER H., HESINA G., GRÖLLER E .: Visual analytics and rendering for tunnel crack analysis. The Visual Computer 32, 6-8 (2016), 859-869. 21
- [OSS*17] ORTNER T., SORGER J., STEINLECHNER H., HESINA G., PIRINGER H., GROLLER E.: Vis-A-Ware: Integrating spatial and nonspatial visualization for visibility-aware urban planning. IEEE Trans. on Visualization and Computer Graphics 23, 2 (2017), 1139-1151. 8, 9, 12, 14. 16. 17. 18. 19

© 2024 The Authors

- [PRd19] PACIFICI M., RAMA F., DE CASTRO MARINS K. R.: Analysis of temperature variability within outdoor urban spaces at multiple scales. *Urban Climate* 27 (2019), 90–104. 14
- [PSS*21] PANDEY A., SYEDA U., SHAH C., GUERRA-GOMEZ J., BORKIN M.: A state-of-the-art survey of tasks for tree design and evaluation with a curated task dataset. *IEEE Trans. on Visualization and Computer Graphics* 28, 10 (2021), 3563–3584. 2
- [PSTB21] PALLIWAL A., SONG S., TAN H., BILJECKI F.: 3D city models for urban farming site identification in buildings. *Computers, Envi*ronment and Urban Systems 86 (2021), 101584. 6, 11, 12, 14, 16, 18
- [PSY*20] PANDEY A., SHUKLA H., YOUNG G. S., QIN L., ZAMANI A. A., HSU L., HUANG R., DUNNE C., BORKIN M. A.: Cerebrovis: Designing an abstract yet spatially contextualized cerebral artery network visualization. *IEEE Trans. on Visualization and Computer Graphics 26*, 1 (2020), 938–948. 21
- [PTD13] PASEWALDT S., TRAPP M., DÖLLNER J.: Multi-perspective detail+overview visualization for 3D building exploration. *Theory and Practice of Computer Graphics* (2013). 7, 12, 17, 18
- [PZA*17] PERHAC J., ZENG W., ASADA S., ARISONA S. M., SCHU-BIGER S., BURKHARD R., KLEIN B.: Urban fusion: Visualizing urban data fused with social feeds via a game engine. In *Proc. of the 2017* 21st International Conference Information Visualisation (2017), IV'17, pp. 312–317. 19

[qgi] QGIS. https://qgis.org. Accessed: May 25, 2023. 17

- [RBC*18] RAMASWAMI A., BETTENCOURT L., CLARENS A., DAS S., ET AL.: Sustainable urban systems: Articulating a long-term convergence research agenda. A Report by the Advisory Committee for Environmental Research and Education (2018). 22
- [RcH*22] RINK K., ŞEN O. O., HANNEMANN M., KÖDEL U., NIX-DORF E., WEBER U., WERBAN U., SCHRÖN M., KALBACHER T., KOLDITZ O.: An environmental exploration system for visual scenario analysis of regional hydro-meteorological systems. *Computers & Graphics 103* (2022), 192–200. 2
- [Rei15] REINHARD K.: Urban design synthesis for building layouts based on evolutionary many-criteria optimization. *International Jour*nal of Architectural Computing 13, 3-4 (2015), 257–269. 8, 9
- [RGN12] RANJBAR H. R., GHARAGOZLOU A. R., NEJAD A. R. V.: 3D analysis and investigation of traffic noise impact from Hemmat highway located in Tehran on buildings and surrounding areas. *Journal of Geographic Information System* 4 (2012). 16, 18
- [RRM*22] ROBINEAU T., RODLER A., MORILLE B., RAMIER D., SAGE J., MUSY M., GRAFFIN V., BERTHIER E.: Coupling hydrological and microclimate models to simulate evapotranspiration from urban green areas and air temperature at the district scale. Urban Climate 44 (2022), 101179. 18
- [RS19] RESCH B., SZELL M.: Human-centric data science for urban studies. *ISPRS International Journal of Geo-Information* 8, 12 (2019), 584. 22
- [RTCVBS17] REDWEIK P., TEVES-COSTA P., VILAS-BOAS I., SAN-TOS T.: 3D city models as a visual support tool for the analysis of buildings seismic vulnerability: The case of Lisbon. *International Journal of Disaster Risk Science* 8, 3 (2017), 308–325. 2
- [RWF*13] RIBICIC H., WASER J., FUCHS R., BLOSCHL G., GROLLER E.: Visual analysis and steering of flooding simulations. *IEEE Trans. on Visualization and Computer Graphics 19*, 6 (2013), 1062–1075. 7, 14, 15, 16, 17, 19
- [SAH20] SHAREEF S., ABU-HIJLEH B.: The effect of building height diversity on outdoor microclimate conditions in hot climate. a case study of Dubai-UAE. Urban Climate 32 (2020), 100611. 18
- [SAL*20] SEIPEL S., ANDRÉE M., LARSSON K., PAASCH J., PAULS-SON J.: Visualization of 3D property data and assessment of the impact of rendering attributes. *Journal of Geovisualization and Spatial Analysis* 4, 2 (2020), 8, 11, 18, 19, 20

© 2024 The Authors.

- [SAYJ18] SHAERI J., AFLAKI A., YAGHOUBI M., JANALIZADEH H.: Investigation of passive design strategies in a traditional urban neighborhood: A case study. Urban Climate 26 (2018), 31–50. 18
- [SCF20] SARETTA E., CAPUTO P., FRONTINI F.: An integrated 3D GIS-based method for estimating the urban potential of BIPV retrofit of façades. Sustainable Cities and Society 62 (2020), 102410. 6, 8
- [SD23] SUN Y., DOGAN T.: Generative methods for urban design and rapid solution space exploration. *Environment and Planning B: Urban Analytics and City Science* 50, 6 (2023), 1577–1590. 8, 16
- [SDP20] SHIRAZ M., DILIMULATI A., PARASCHIVOIU M.: Wind power potential assessment of roof mounted wind turbines in cities. *Sustainable Cities and Society* 53 (2020), 101905. 7, 9, 12, 16
- [SH20] SCHROTTER G., HÜRZELER C.: The digital twin of the city of Zurich for urban planning. *Journal of Photogrammetry, Remote Sensing* and Geoinformation Science 88, 1 (2020), 99–112. 2
- [SJ19] ŚLUSARSKI M., JURKIEWICZ M.: Visualisation of spatial data uncertainty. A case study of a database of topographic objects. *ISPRS International Journal of Geo-Information* 9, 1 (2019), 16. 20
- [SKB*13] SHOJAEI D., KALANTARI M., BISHOP I., RAJABIFARD A., AIEN A.: Visualization requirements for 3D cadastral systems. *Computers, Environment and Urban Systems* 41 (2013), 39–54. 8, 11, 14, 19
- [SKK08] STOTER J., KLUIJVER H., KURAKULA V.: 3D noise mapping in urban areas. *International Journal of Geographical Information Sci*ence 22, 8 (2008), 907–924. 8, 9
- [SL19] SHI Y., LI X.: Evaluating the influence of three simplifications on natural ventilation rate simulation. In *Proc. of the 2019 Symposium on Simulation for Architecture and Urban Design* (San Diego, USA, 2019), SimAUD '19, pp. 1–4. 7
- [SMS23] SALIMI S., MIRGHOLAMI M., SHAKIBAMANESH A.: Visual impact assessment of urban developments around heritage landmarks using ULVIA method: (the case of Ark-e-Alishah monument in Tabriz). *Environment and Planning B: Urban Analytics and City Science 50*, 3 (2023), 678–693. 8, 18
- [SOP*15] SORGER J., ORTNER T., PIRINGER H., HESINA G., GROLLER E.: A taxonomy of integration techniques for spatial and nonspatial visualizations. In *Vision, Modeling & Visualization* (2015), The Eurographics Association. 15
- [SORB18] SHOJAEI D., OLFAT H., RAJABIFARD A., BRIFFA M.: Design and development of a 3d digital cadastre visualization prototype. *ISPRS International Journal of Geo-Information* 7, 10 (2018). 8
- [SVS*22] SRABANTI S., VEIGA C., SILVA E., LAGE M., FERREIRA N., MIRANDA F.: A comparative study of methods for the visualization of probability distributions of geographical data. *Multimodal Technologies and Interaction* 6, 7 (2022), 53. 20
- [SWZ21] SUN G., WEBSTER C., ZHANG X.: Connecting the city: A three-dimensional pedestrian network of Hong Kong. *Environment and Planning B: Urban Analytics and City Science* 48, 1 (2021), 60–75. 8
- [TCFR21] TURAN I., CHEGUT A., FINK D., REINHART C.: Development of view potential metrics and the financial impact of views on office rents. *Landscape and Urban Planning 215* (2021), 104193. 8, 12
- [TLH*22] TANG J.-H., LIN B.-C., HWANG J.-S., CHEN L.-J., WU B.-S., JIAN H.-L., LEE Y.-T., CHAN T.-C.: Dynamic modeling for noise mapping in urban areas. *Environmental Impact Assessment Review 97* (2022), 106864. 8, 9, 12, 15
- [TZQ*19] TIAN Y., ZHOU W., QIAN Y., ZHENG Z., YAN J.: The effect of urban 2D and 3D morphology on air temperature in residential neighborhoods. *Landscape Ecology* 34, 5 (2019), 1161–1178. 2
- [VDP*15] VALDIVIA P., DIAS F., PETRONETTO F., SILVA C., NONATO L.: Wavelet-based visualization of time-varying data on graphs. In Proc. of the 2015 IEEE Conf. on Visual Analytics Science and Technology (Chicago, USA, 2015), VAST '15, pp. 1–8. 21

- [VGDA*12] VANEGAS C., GARCIA-DORADO I., ALIAGA D., BENES B., WADDELL P.: Inverse design of urban procedural models. ACM Trans. on Graphics 31, 6 (2012), 1–11. 7, 15, 19
- [VJP*21] VIRTANEN J.-P., JAALAMA K., PUUSTINEN T., JULIN A., HYYPPÄ J., HYYPPÄ H.: Near real-time semantic view analysis of 3D city models in web browser. *ISPRS International Journal of Geo-Information 10*, 3 (2021), 138. 8
- [VSOC21] VUCKOVIC M., SCHMIDT J., ORTNER T., CORNEL D.: Combining 2D and 3D visualization with visual analytics in the environmental domain. *Information 13*, 1 (2021). 7, 10, 16, 17
- [Wan17] WANG Y.: Deck.gl: Large-scale web-based visual analytics made easy. In *The IEEE Workshop on Visualization in Practice* (2017). 17
- [WBH14] WILKINSON S., BRADBURY G., HANNA S.: Approximating urban wind interference. In Proc. of the 2014 Symposium on Simulation for Architecture and Urban Design (San Diego, USA, 2014), SimAUD '14, pp. 1–8. 7, 9, 15, 17
- [WDDH19] WILSON L., DANFORTH J., DAVILA C., HARVEY D.: How to generate a thousand master plans: A framework for computational urban design. In *Proc. of the 2019 Symposium on Simulation for Architecture and Urban Design* (San Diego, USA, 2019), SimAUD '19, pp. 1–8. 6, 16, 17, 19
- [WFC*22] WANG C., FERRANDO M., CAUSONE F., JIN X., ZHOU X., SHI X.: Data acquisition for urban building energy modeling: A review. *Building and Environment 217* (2022), 109056. 21
- [WHG10] WU H., HE Z., GONG J.: A virtual globe-based 3D visualization and interactive framework for public participation in urban planning processes. *Computers, Environment and Urban Systems 34*, 4 (2010), 291–298. 6, 8, 12
- [WHM*19] WANG C., HOU J., MILLER D., BROWN I., JIANG Y.: Flood risk management in sponge cities: The role of integrated simulation and 3D visualization. *International Journal of Disaster Risk Reduction* 39 (2019), 101139. 2
- [WHN*21] WONG N. H., HE Y., NGUYEN N. S., RAGHAVAN S. V., MARTIN M., HII D. J. C., YU Z., DENG J.: An integrated multiscale urban microclimate model for the urban thermal environment. *Urban Climate 35* (2021), 100730. 7, 11
- [WKS*14] WASER J., KONEV A., SADRANSKY B., HORVÁTH Z., RIBIČIĆ H., CARNECKY R., KLUDING P., SCHINDLER B.: Many plans: Multidimensional ensembles for visual decision support in flood management: Many plans. *Computer Graphics Forum 33*, 3 (2014), 281– 290. 6, 7, 10, 12, 15, 19
- [WM20] WOLOSIUK D., MAHDAVI A.: Application of ontologically structured data for building performance analysis. In Proc. of the 2020 Symposium on Simulation for Architecture and Urban Design (San Diego, USA, 2020), SimAUD '20, pp. 1–6. 6, 8, 15
- [WP16] WU M.-L., POPESCU V.: Multiperspective focus+context visualization. *IEEE Trans. on Visualization and Computer Graphics* 22, 5 (2016), 1555–1567. 9, 18
- [WPH17] WANG C., POULIOT J., HUBERT F.: How users perceive transparency in the 3D visualization of cadastre: testing its usability in an online questionnaire. *GeoInformatica* 21, 3 (2017), 599–618. 8
- [WSK18] WILLENBORG B., SINDRAM M., KOLBE T. H.: Applications of 3D city models for a better understanding of the built environment. *Trends in Spatial Analysis and Modelling: Decision-Support and Planning Strategies* (2018), 167–191. 6, 7, 8, 17, 18
- [WUAC23] WICKRAMATHILAKA N., UJANG U., AZRI S., CHOON T. L.: Three-dimensional visualisation of traffic noise based on the Henk de-Klujijver model. *Noise Mapping 10*, 1 (2023), 20220170. 9, 16
- [WYX*22] WEN J., YANG S., XIE Y., YU J., LIN B.: A fast calculation tool for assessing the shading effect of surrounding buildings on window transmitted solar radiation energy. *Sustainable Cities and Society 81* (2022), 103834. 20

- [XGY*23] XU Y., GUO C., YANG J., YUAN Z., HO H.: Modelling impact of high-rise, high-density built environment on COVID-19 risks: Empirical results from a case study of two Chinese cities. *International Journal of Environmental Research and Public Health 20*, 2 (2023), 1422. 2
- [XLH*15] XIONG C., LU X., HORI M., GUAN H., XU Z.: Building seismic response and visualization using 3D urban polygonal modeling. *Automation in Construction* 55 (2015), 25–34. 7, 12, 15
- [YAM*20] YUAN C., ADELIA A. S., MEI S., HE W., LI X.-X., NOR-FORD L.: Mitigating intensity of urban heat island by better understanding on urban morphology and anthropogenic heat dispersion. *Building* and Environment 176 (2020), 106876. 2
- [YCL*19] YING S., CHEN N., LI W., LI C., GUO R.: Distortion visualization techniques for 3D coherent sets: A case study of 3D building property units. *Computers, Environment and Urban Systems* 78 (2019), 101382. 8, 14, 18
- [YJB22] YAP W., JANSSEN P., BILJECKI F.: Free and open source urbanism: Software for urban planning practice. *Computers, Environment* and Urban Systems 96 (2022), 101825. 22
- [YJNY11] YASUMOTO S., JONES A., NAKAYA T., YANO K.: The use of a virtual city model for assessing equity in access to views. *Computers, Environment and Urban Systems* 35, 6 (2011), 464–473. 8
- [YNK*18] YAO Z., NAGEL C., KUNDE F., HUDRA G., WILLKOMM P., DONAUBAUER A., ADOLPHI T., KOLBE T.: 3DCityDB - a 3D geodatabase solution for the management, analysis, and visualization of semantic 3D city models based on CityGML. *Open Geospatial Data, Software and Standards* 3, 1 (2018). 6, 9, 15
- [YYS*16] YU S., YU B., SONG W., WU B., ZHOU J., HUANG Y., WU J., ZHAO F., MAO W.: View-based greenery: A three-dimensional assessment of city buildings' green visibility using Floor Green View Index. Landscape and Urban Planning 152 (2016), 13–26. 8, 9, 10, 12, 14
- [ZC22] ZHANG L., CHIARADIA A.: Walking in the cities without ground, how 3D complex network volumetrics improve analysis. *Envi*ronment and Planning B: Urban Analytics and City Science 49, 7 (2022), 1857–1874. 2
- [Zha13] ZHANG J.: A study of the relationship between urban form and environmental performance for three urban block typologies in Paris. In *Proc. of the 2013 Symposium on Simulation for Architecture and Urban Design* (San Diego, USA, 2013), SimAUD '13, pp. 1–10. 6, 12
- [ZKL*21] ZHANG S., KWOK K., LIU H., JIANG Y., DONG K., WANG B.: A CFD study of wind assessment in urban topology with complex wind flow. Sustainable Cities and Society 71 (2021), 103006. 7, 12, 15, 16, 17, 20
- [ZLH*17] ZHANG J., LAU S., HENG C., LAU S.-K., LAI H.: A case study on the relationship between urban morphology and traffic noise distribution in high-density urban context. In *Proc. of the 2017 Symposium on Simulation for Architecture and Urban Design* (San Diego, USA, 2017), SimAUD '17, pp. 1–7. 8, 15, 16
- [ZSW21] ZHAO J., SUN G., WEBSTER C.: Walkability scoring: Why and how does a three-dimensional pedestrian network matter? *Environment and Planning B: Urban Analytics and City Science* 48, 8 (2021), 2418–2435. 8
- [ZWC*16] ZHENG Y., WU W., CHEN Y., QU H., NI L. M.: Visual analytics in urban computing: An overview. *IEEE Trans. on Big Data* 2, 3 (2016), 276–296. 3
- [ZWK*22] ZHU R., WONG M. S., KWAN M.-P., CHEN M., SANTI P., RATTI C.: An economically feasible optimization of photovoltaic provision using real electricity demand: A case study in new york city. *Sustainable Cities and Society* 78 (2022), 103614. 6, 8, 12
- [ZWL*23] ZHENG H., WU B., LIN H., JIA J., WEI H.: Feasibility assessment of solar photovoltaic deployments on building surfaces with the constraint of visual impacts. *Environment and Planning B: Urban Analytics and City Science 50*, 6 (2023), 1591–1606. 6

© 2024 The Authors. Computer Graphics Forum published by Eurographics and John Wiley & Sons Ltd.

- [ZWLZ22] ZHU X., WANG X., LEI L., ZHAO Y.: The influence of roadside green belts and street canyon aspect ratios on air pollution dispersion and personal exposure. Urban Climate 44 (2022), 101236. 7
- [ZXL22] ZHANG A., XIA C., LI W.: Exploring the effects of 3D urban form on urban air quality: Evidence from fifteen megacities in China. *Sustainable Cities and Society* 78 (2022), 103649. 2
- [ZY18] ZENG W., YE Y.: VitalVizor: A visual analytics system for studying urban vitality. *IEEE Computer Graphics and Applications 38*, 5 (2018), 38–53. 8, 9, 11, 16, 17, 19
- [ZYS*19] ZHU R., YOU L., SANTI P., WONG M., RATTI C.: Solar accessibility in developing cities: A case study in Kowloon East, Hong Kong. Sustainable Cities and Society 51 (2019), 101738. 6
- [ZZL21] ZHANG C., ZENG W., LIU L.: UrbanVR: An immersive analytics system for context-aware urban design. *Computers & Graphics 99* (2021), 128–138. 6, 8, 9, 12, 16, 17, 18, 19, 20